## CHAPTER 13 - - The Wealth Gap

## I. Introduction

Wealth—what you own minus what you owe—is the key to economic security.<sup>1</sup> It is what enables families to build a better future.<sup>2</sup> Wealth functions in many ways. It provides economic stability during lean times. It opens doors to improving quality of life. It is a dam against the floodwaters of economic catastrophes. It provides access to political power and it allows us to live and retire with dignity.

Wealth can also be passed down through generations, allowing children to live better lives than their parents and grandparents. It allows parents to pay for their children's college education. It allows grandparents to help a young family buy their first home. Throughout American history, government policies at all levels have helped white families collect these tools while preventing or undermining African American families' ability to do the same.

As a result, the wealth gap between African American and white Americans is enormous and roughly the same today as it had been two years before the Civil Rights Act was passed in 1964.<sup>3</sup> In 2019, the median African American household had a net worth of \$24,100, less than 13 percent of the median net worth of white households at \$188,200.<sup>4</sup>

This wealth gap persists regardless of education level and family structure.<sup>5</sup> For example, at the median, single African American women over the age of 60 with a college degree— at \$11,000—have less than three percent of the wealth of single white women over the age of 60 with a college degree— at \$384,000.<sup>6</sup> Single white parents have more than double the wealth (\$35,000), at the median, than married African American parents (\$16,000).<sup>7</sup>

The wealth gap is present across all income levels. In 2016, estimates drawn from the Survey of Consumer Finances indicate the median white household in the bottom fifth of incomes, or the poorest "quintile" of white households, had a net worth of \$21,700, which is greater than the median net worth of \$18,601 for *all* Black households.<sup>8</sup> Black households in the bottom fifth of incomes had a median net worth of \$2,700, less than one eighth as much as the poorest quintile of white households.<sup>9</sup>



The trend is the same across social classes. In 2019, the median white working-class household had a net worth of \$114,270, while the median African American professional-managerial household had a net worth of \$38,800. In the same year, white professional-managerial households—at \$276,000—had a median net worth that was eight times the median African American professional-managerial household and *19 times* the median African American working-class household.<sup>10</sup>

This wealth gap is the result of the discrimination that African Americans experience, as described in the previous chapters.<sup>11</sup> The American government at the federal, state, and local levels has systematically prevented African American communities from building, maintaining, and passing on wealth. These harms cascade over a lifetime and compound over generations. support of racial terror, disenfranchisement, land theft, mass incarceration, exclusion of African Americans from government benefits, and banking discrimination. Unequal homeownership, fewer assets, and lower business ownership continue to drive the wealth gap today. This has resulted in racial differences in the capacity of African Americans to transmit resources across generations, lower financial resilience during crises, and homelessness.

Section II discusses estimates of the contemporary racial wealth gap for the nation as whole, for California, by gender, and for descendants of Africans enslaved in the United States. Section III describes historical causes of the racial wealth gap during enslavement and post-enslavement, including racial terror, land theft, mass incarceration, and discrimination in government bene-

> fits, the labor market, and banking. Section IV discusses the drivers of the contemporary wealth gap today include unequal homeownership, fewer assets, and lower business ownership. Section V discusses the harmful effects of the wealth gap,

The wealth gap was roughly the same in 2016 as it was in 1962.

The historical causes of the wealth gap is based in enslavement and legal segregation and continues through ongoing racial inequality and racism today. They include direct government creation of white wealth and destruction of African American wealth through the which has resulted in racial differences in the capacity of African Americans to transmit resources across generations, lower financial resilience during crisis, and increased homelessness.

## II. The Contemporary Racial Wealth Gap

#### **National and California Estimates**

Significant research demonstrate that white Americans have long had a higher net worth than African Americans. The gap has changed little since 1989, when the median white household wealth was \$143,000, and the median African American household wealth was \$9,000, approximately 94 percent less than white household wealth.<sup>12</sup> The wealth gap was roughly the same in 2016 as it was in 1962,<sup>13</sup> two years before the Civil Rights Act. Preliminary research suggests that, despite rapid accumulation of wealth by African Americans in the decades after slavery and a narrowing of the racial wealth gap during World War II and the Civil Rights era, this progress halted by the mid-20th century with the racial wealth gap widening over the last several decades.<sup>14</sup> From 2005 to 2019, an interval that captures much of the impact of the Great Recession, median household wealth -all assets minus all debt-among African Americans fell 53 percent, compared with a drop of 16 percent among white Americans.<sup>15</sup>

An asset is anything you own that adds financial value, as opposed to a liability, which is money you owe.<sup>16</sup> Examples of personal assets include: a home or other property, such as a rental house or commercial property; a checking or savings account; cars; financial and retirement accounts; gold, jewelry, and coins; collectibles and art; and life insurance policies.<sup>17</sup>

Wealth estimates can be demonstrated in median and mean figures, both of which are provided in this chapter. A median figure shows the worth of the middle household in each community.<sup>18</sup> A mean figure shows the worth of the average household in the community.<sup>19</sup> Some researchers suggest that the median is a more useful measure for estimating differences in wealth between African American and white people because it is not affected by exceptions like the few extremely rich individuals who would skew the average higher than is representative.<sup>20</sup> However, researchers also suggest that the mean is the appropriate target measure for calculating the sum required to eliminate the racial wealth gap.<sup>21</sup>

Policymakers have usually focused on the median gap in wealth, which some researchers argue is not representative of what is happening to the average African American or white person in reality.<sup>22</sup> Comparing African American and white wealth at the mean—for the average household in each community—shows a far larger wealth gap.<sup>23</sup> Differences in family structure also do not explain the racial wealth gap.<sup>33</sup> Single white women with children have a higher median net worth than African American women with no children.<sup>34</sup> At the median, single white parents have more than two times the wealth (\$35,000) than married African American parents (\$16,000).<sup>35</sup>

Today, white American households continue to be far more likely to hold assets, and the types of assets they hold are worth, on average, more than that of African American households.<sup>24</sup> In 2019, the most recent year for which data is available, the total financial assets of white households is nine times higher than African American households.<sup>25</sup> The median African American household wealth was approximately \$24,100, while the median white household wealth was approximately \$188,200–a dif-

This wealth disparity cannot be explained by lack of personal motivation and effort, family instability, or lack of education. For example, in 2019, black professional-managerial households had a net worth of \$38,800, while white professional-managerial households had a net worth of \$276,000. Single white parents had more than two times the wealth— at \$35,000 —of married black parents— at \$16,000.

ference of \$164,100.<sup>26</sup> The mean for African American household wealth is \$142,300, while the mean for white household wealth is close to \$1 million at \$983,400-a difference of \$840,900.<sup>27</sup>

#### In 2019, white households owned



#### assets than African American households

This wealth disparity cannot be explained by lack of personal motivation and effort, lack of financial literacy, family instability, lack of education, lack of homeownership, or lack of entrepreneurship on the part of African Americans.<sup>28</sup> For instance, there is very little, if any, evidence to support the claim that African American saving behavior is the source of the enormous racial wealth gap.<sup>29</sup> There is no significant difference in savings between African American and white families with similar income levels, nor a difference in rates of return on their personal investments.<sup>30</sup> In fact, in some income categories, African American people display a higher rate of savings.<sup>31</sup> And though African Americans have more family obligations on their income than white Americans because their relatives are more likely to need help than white Americans, the savings rate among African American people is comparable to the savings rate at each level of household income among white people.32

Nor does effort or education. For comparable levels of family socioeconomic status, Black youth obtain more years of schooling and credentials, including college degrees, than white youth.<sup>36</sup> And the wealth gap exists between African American and white women regardless of whether or not they have a bachelor's degree.<sup>37</sup>

Although the wealth gap and its causes in California and the nation is an under-studied area, preliminary studies suggest that the racial wealth gap in California is the same or worse than it is at the national level.

Some studies extrapolate California's racial wealth gap from national estimates.<sup>38</sup> Direct California estimates of the racial wealth gap are only available for a single metropolitan area: Los Angeles.<sup>39</sup> In 2016, while the median net worth of white Angelino households (assets minus debts) was \$355,000, median net worth of native-born African American Angelino households was \$4,000.<sup>40</sup> The average African American household in Los Angeles had only just had one percent of the median wealth of the average white household, far worse than the national average of 10 to 15 percent.<sup>41</sup>

#### **Gender-Specific Issues**

The wealth gap between African American men and African American women, which is small,<sup>42</sup> functions differently than the wealth gap between white men and white women, which is much larger. African American men and women have similarly low wealth, although for slightly different reasons.<sup>43</sup>

The wealth gap between white men and women is largely because white men have traditionally had access to jobs that provide retirement accounts and other benefits available to careers not available to women.<sup>44</sup> This difference in access does not exist between African American men and women, as African Americans of all genders historically have been excluded from these benefits due labor discrimination, as discussed in Chapter 10, Stolen Labor and Hindered Opportunity.

The median wealth for single African American women is \$200, while the median wealth of single African American men is \$300.<sup>45</sup> Studies that show a greater wealth gap between Black women and men do not appear to take into account incarcerated Black men, who are deprived of their ability to build wealth for themselves and their family during a prison sentence.<sup>46</sup>

#### **2015 MEDIAN WEALTH FOR SINGLES**



Federal statistical agencies, including the U.S. Census Bureau, collect little information about people who are incarcerated, and they are excluded from household samples in national surveys.<sup>47</sup> As a result, these individuals are invisible to most mainstream social institutions, lawmakers, and nearly all social science research not directly related to crime or criminal justice.<sup>48</sup> And since African Americans are six times more likely to be incarcerated than white Americans, this has the effect of making it appear that African Americans are better off financially than they really are.<sup>49</sup> As discussed below in section III, criminal convictions also create numerous barriers to wealth building even after the sentence ends. Still, African American women face barriers to wealth building due to the combination of racism and sexism. As a result, there are vast differences in wealth between African American women and both white men and white women.<sup>50</sup> One study reports that, in 2019, single Black women's median net wealth was \$7,000, while median net wealth for white women was \$85,000 and \$92,000 for single white men.<sup>51</sup> While white men's median wealth was \$28,900 in 2015, African American women's median wealth was \$200, less than one cent on every dollar of white men's median wealth.<sup>52</sup>

The large wealth gap faced by single Black women is particularly important because Black women are more and increasingly likely to be single and breadwinners.<sup>53</sup> The marriage rate of Americans aged 25 to 54 has declined since the early 1960s across all groups, but especially for Black women for whom it has halved to less than 40 percent.<sup>54</sup> One-third of Black women aged 25 to 54 are single with children in the household.<sup>55</sup> Among Black mothers, more than 80 percent are breadwinners compared to 50 percent of white mothers.<sup>56</sup> At least half of unmarried Black women have zero or negative assets.<sup>57</sup> On average, Black women do not accumulate net worth approaching retirement; they have no financial assets at age 50 and do not accumulate any more as they age.<sup>58</sup>

African American men also face a stark wealth gap with white women and men. One study reports that, in 2015, median net wealth for African American men was 15 percent of the median net wealth for white men and 16 percent of the wealth of white women.<sup>59</sup> While white men's median wealth was \$28,900 in 2015, African American men's median was \$300, about one cent on every dollar of white men's wealth.<sup>60</sup>

There does not appear to be extensive research on wealth gap estimates for African American LGBTQ populations.

## Estimates Based on Immigration and Migration Patterns

Today, approximately 41.1 million African American people live in the United States, according to the 2020 census.<sup>61</sup> Of those 41.1 million African American individuals, experts differ on how many are the descendants of African American people enslaved in the United States.

About 12 percent of Black people in America were born in a foreign country.<sup>62</sup> Nine percent of African Americans have at least one foreign-born parent.<sup>63</sup> By 2060, the Black foreign-born population is projected to make up about one-third of the U.S. Black population.<sup>64</sup> Fiftyeight percent of Black immigrants arrived in the United States after 2000.<sup>65</sup> Of the current 2.8 million African American Californians, 244,969 are foreign born, according to the U.S. Census Bureau.<sup>66</sup>

Every U.S. census conducted since 1970 has found that African American immigrants from the English speaking Caribbean earn more, are more likely to be employed than U.S.-born African Americans, are more likely to hold more financial assets,<sup>67</sup> are more likely to own their home,<sup>68</sup> and most are more likely to be healthy than U.S.-born African Americans.<sup>69</sup>

There appears to be no data at the national and state level and limited scholarship at the city level describing the wealth gap between descendants of African American people enslaved in the United States, recent African American immigrants, and white Americans.<sup>70</sup> Very few of the city level studies present findings on the wealth gap that disaggregates the racial category of African American by national origin.<sup>71</sup> Some scholars argue that the effects of systemic racism have uniquely harmed African American descendants of slavery when compared to African American immigrants who do not have the same experience of systemic racial discrimination in the United States.<sup>72</sup>

One study, "*The Color of Wealth in Los Angeles*," included separate data for U.S. born African Americans and recent immigrants from Africa.<sup>73</sup> National origin and race were both self-reported in this study. <sup>74</sup> On average, white Angelinos were far more likely to hold assets in stocks, mutual funds, and investment trusts than both U.S.-born and African Black Angelinos.<sup>75</sup>

But, African Black Angelinos were likely to hold more assets than U.S.-born Black Angelinos. For example, 87 percent of African Black Angelinos owned liquid assets versus 62 percent of Black Angelinos born in the United States.<sup>76</sup> Eighty percent of African Black Angelinos had a checking account versus 68 percent of U.S.-born African Americans.<sup>77</sup> Eighty percent of African Black Angelinos had a savings account versus 56 percent of U.S.-born Black Angelinos.<sup>78</sup> Such studies in other cities reflect comparable wealth gaps between U.S.-born African Americans and recent immigrants.<sup>79</sup>

## III. Historical Causes of the Racial Wealth Gap

The modern racial wealth gap between African Americans and other racial groups began with enslavement. But scholars debate whether enslavement should be the basis for reparations given that today's wealth gap is the cumulative effects of racism over centuries.<sup>80</sup> This section describes historical causes of the racial wealth gap during and post-slavery, including racial terror, land theft, mass incarceration, and discrimination in government benefits, the labor market, and banking.

#### Enslavement

Several scholars have estimated the slavery bill for reparations.<sup>81</sup> Most of these estimates require a calculation in today's dollars for unpaid wages, the purchase prices of the human property, or the land promised to the formerly enslaved.<sup>82</sup> These estimates are generated by multiplying earlier values by a compounding interest rate.<sup>83</sup> For example, Thomas Craemer's calculations for unpaid wages owed to enslaved people amounts to \$19.4 trillion in today's dollars.<sup>84</sup> He arrives at this number by multiplying the prevailing average market wage by the number of hours worked for each 24-hour day by those enslaved over the interval of 1776 to 1865 and applies a three percent interest rate.<sup>85</sup> Merely doubling the interest rate to the more realistic six percent would increase the total estimate to \$6.6 quadrillion in 2019 dollars.  $^{86}$ 

Similar to Craemer's estimates is James Marketti's bill using the idea of income diverted from enslaved persons, arriving at a figure of \$2.1 trillion as of 1983.<sup>87</sup> Using a six percent interest rate; the 2018 value amounts to \$17.1 trillion.<sup>88</sup> Other estimates are reached by calculating the value in today's dollars of the wealth held in enslaved

COURTESY OF LIBRARY OF CONGRESS



White homesteaders in El Monte. The federal government essentially gave land away for free to mostly white families through the Homestead Act. (1936)

people as property. For example, Judah P. Benjamin, a critical member of Jefferson Davis's Confederate Cabinet proclaimed the value of enslaved persons in 1860 to be four billion dollars, which compounds to \$42.2 trillion by 2019 at a six percent interest rate.<sup>89</sup> Other scholars argue that these large sums are underestimates because they do not account for the physical and emotional harms of enslavement, the coercive nature of the system, the denial of the ability to acquire property, or the deprivation of autonomy.<sup>90</sup>

#### **Post-Enslavement**

The case for reparation extends beyond slavery to the near-century-long era of legal segregation, violence and terror, and the atrocities that continue today: mass incarceration; police killings of unarmed African Americans; sustained credit, housing, employment discrimination; and the immense Black-white wealth disparity.<sup>91</sup> Scholars have divided post-slavery American history into five overlapping phases of federal government policies, which created the modern racial wealth gap. They involve white wealth creation through government land grants and mortgage subsidies and African American wealth destruction through racial terror, eminent domain, and mass incarceration.

Between 1868 and 1934, the federal government gave 246 million acres of land essentially for free to mostly white Americans—an area close to the size of California and Texas combined. More than 1.5 million white families received land patents, and today as many as 46 million of their living descendants reap the wealth benefits, approximately one-quarter of the adult population of the United States.

## White Wealth Creation through Government Land Grants

From 1862 to 1976, the federal government transferred massive amounts of land mostly to white families. Some scholars have named this phase the Wagon Train phase, after the covered-wagon caravans romanticized by 1950s television shows, which carried white families to seek their fortunes in the West.<sup>92</sup>

In 1862, the federal government established the Homestead Act, which distributed land until about 1980, although more occasionally after the 1920s.<sup>93</sup> The Homestead Act encouraged western migration by providing American citizens—and immigrants soon to be citizens—up to 160

acres of public lands, which was increased to 320 acres in 1909—for 0 if they continuously lived on the property for five years and paid a small 0 filing fee.

Homesteaders also had the option of paying \$1.25 per acre if they lived on the property for six months.<sup>95</sup> While the language of the act did not exclude people based on race, African Americans were unable to secure land allocations under the act for four years until the Civil Rights Act of 1866 clarified that they were citizens.<sup>96</sup> California's homestead laws similarly excluded African Americans before 1900 because they required a homesteader to be a white citizen.<sup>97</sup>

Though African American homesteaders were able to secure land allocations under the Act after 1866, they were few in comparison to the multitudes of white settlers and had to settle for the least desirable land.<sup>98</sup> Between 1868 and 1934, the federal government gave 246 million acres of land essentially for free to mostly white, native-born and immigrant Americans—an area close to the size of California and Texas combined.<sup>99</sup> More than 1.5 million white families received land patents, and today as many as 46 million of their living descendants reap the wealth benefits, approximately one-quarter of the adult population of the United States.<sup>100</sup> In comparison and as an example, approximate-

> ly 3,500 African American claimants succeeded in obtaining their patents from the General Land Office in the Great Plains, granting them ownership of approximately 650,000 acres of prairie land.<sup>101</sup>

> The federal government undermined other efforts to allocate land to the formerly enslaved. Another estimate of reparations to African Americans can be made by calculating the value in today's dollars of the unfulfilled land distribution of "forty acres and a mule" promised to

the formerly enslaved beginning in 1865.<sup>102</sup> On January 16, 1865, upon seizing the coastline from Charleston, South Carolina to St. John's River, Florida,<sup>103</sup> General Sherman issued Special Field Orders No. 15 that established the provision "of not more than (40) forty acres of tillable ground" designated "for the settlement of the negroes now made free by the acts of war and the proclamation of the President of the United States."<sup>104</sup>

The order carved out 400,000 acres of land confiscated or abandoned by Confederates. Each family of formerly enslaved African American people would get up to 40 acres.<sup>105</sup> The Army would lend them mules no longer in use.<sup>106</sup> Further, the Freedmen's Bureau Act of March 3, 1865, had an explicit racial land redistribution provision that specified that "not more than 40 acres" of land was to be provided to refugee or freed male citizens at three years' annual rent not exceeding six percent of the value of the land based on appraisal of the state tax authorities in 1860.<sup>107</sup> At the end of three years of occupying the

## African American Wealth Destruction through Racial Terror

From 1865 to the present, federal, state and local government actors refused to protect African Americans as they faced violence and land theft. Sometimes, government actors joined, led, or supported the violence.<sup>117</sup> As

General William Tecumseh Sherman's famous 40 acres and a mule field order, given in 1865, amounts to \$1.3 trillion at six percent interest, in today's dollars. detailed in Chapter 3, Racial Terror, white federal, state, and local government officials, working jointly with private citizens, terrorized African Americans to prevent them from accumulating political and economic power.

land, they could purchase it and receive title.<sup>108</sup> Similar provisions were included in the postwar Southern Homestead Act of 1866. Freedmen were to receive land in the southern states at a price of \$5 for 80 acres.<sup>109</sup>

President Andrew Johnson intensely opposed these acts and neither were effectively implemented on behalf of the formerly enslaved.<sup>110</sup> By the end of 1865, Johnson also had ordered the removal of thousands of formerly enslaved persons from the lands they had settled under Sherman's Special Field Orders No. 15,<sup>111</sup> which were ultimately given back to former enslavers.<sup>112</sup> With the exception of a small number who had legal land titles, freed African American people were removed from the land as a result of President Johnson's "restoration" program.<sup>113</sup>

If four million enslaved persons had gained emancipation by 1865, and the land allocation rule meant that roughly 40 acres would go to families of four, each formerly enslaved individual would have been allocated about 10 acres. This implies a total distribution of at least 40 million acres of land.

Using a conservative estimate of \$10 per acre, the total value of the projected distribution of land to the freedmen would have been \$400 million in 1865.<sup>114</sup> The value in today's dollars at a six percent interest rate would be

\$1.3 trillion. This number would be much higher if the conditions of the Southern Homestead Act, which provided for 80 acres of land to be sold to freedman at \$5 total were treated as a debt to be paid to the descendants of the formerly enslaved.<sup>115</sup> If, as some scholars in-

terpret, *each* freedman was eligible to receive 40 acres of land, it would have led to a much higher total value of the land to be distributed to freedmen after the war amounting to \$1.6 billion in 1865 and compounding to \$12.6 trillion at a six percent interest rate in 2019.<sup>116</sup> White mobs destroyed thriving African American communities in racial massacres nationwide in Louisiana, North Carolina, Michigan, Delaware, Nebraska, Florida, Illinois, Oklahoma, Texas, and elsewhere.<sup>118</sup> The most well-known was the destruction of the Greenwood district in Tulsa, Oklahoma. The Greenwood district was known popularly as "Black Wall Street."<sup>119</sup> Scholars estimate that the present value of destroyed African American property in Tulsa is at least \$100 million.<sup>120</sup> The 1919 massacre in Elaine, Arkansas destroyed \$10 million of African American wealth.<sup>121</sup>

Evidence exists that murders of African Americans continue to be driven by underlying economic incentives.<sup>122</sup> Police killings of unarmed African Americans frequently occur in neighborhoods undergoing white gentrification.<sup>123</sup>

#### White Wealth Creation through Mortgage Subsidies

In the 19th century, federal, state, and local governments passed laws and implemented practices that heavily subsidized the creation of the white middle class while substantively crippling the ability of African American people to do the same.<sup>124</sup> Federal policies, implemented by private citizens, focused on helping mostly white Americans buy single-family homes.<sup>125</sup> As discussed in Chapter 5, The Root of Many Evils: Housing

Scholars estimate that the present value of destroyed Black property in Tulsa is at least \$100 million.

Segregation, the Veterans Administration, the Federal Housing Administration, and the Home Owner's Loan Corporation helped white families buy single family homes in the suburbs while preventing African American families from doing the same.<sup>126</sup>

Beginning in the 1930s and 1940s, the federal government created programs that subsidized low-cost loans and opened up home ownership to millions of average Americans for the first time.<sup>127</sup> At the same time, government underwriters introduced a national appraisal system, tying property value and loan eligibility in part to neighborhood racial composition, which designated predominantly nonwhite neighborhoods as hazardous and coloring these areas red—a pro-

cess known as redlining.<sup>128</sup>

renewal, erasing generations of accumulated African American wealth.<sup>134</sup> African American business districts were cleaved by the highways, and never recovered.<sup>135</sup>

In the mid-20<sup>th</sup> century, the United States Department of Agriculture's (USDA) policies discriminated against and displaced African Americans.<sup>136</sup> During the civil rights era, federal anti-discrimination statutes that applied

White communities received the highest ratings and benefited from low-cost, government backed loans. Minority and mixed neighborhoods—and especially African American neighborhoods—received the lowest ratings and were denied these loans.<sup>129</sup> This functionally concentrated African Americans

Funded by the federal government, state and local officials used eminent domain to destroy countless thriving Black communities in the name of highway and park construction, and urban renewal, erasing generations of accumulated Black wealth. African American business districts were cleaved, and never recovered.

into impoverished neighborhoods in America's urban centers.<sup>130</sup> Of the \$120 billion worth of new housing subsidized between 1934 and 1962, less than two percent went to nonwhite families—virtually locking them out of homeownership.<sup>131</sup>

Today, approximately three in four neighborhoods—74 percent—that the federal government deemed "hazard-ous" in the 1930s remain low- to moderate-income, and more than 60 percent are predominantly nonwhite.<sup>132</sup>

# of subsidized new housing went to nonwhite families from 1934 – 1962

## African American Wealth Destruction through Disenfranchisement and Land Theft

As described in Chapter 5, Housing Segregation, the federal government passed the National Highway Act of 1956 and urban renewal legislation.<sup>133</sup> Funded by the federal government, state and local officials used eminent domain to destroy thriving African American communities in the name of highway construction and urban

to the USDA were diluted by the time they reached the local level, and did not provide protection for African American farmers.<sup>137</sup> White USDA administrators gave millions of dollars in funding to all-white Southern local agricultural committees.<sup>138</sup> These powerful committees were county arms of the USDA and did not want African American farmers on their boards, so they would prevent their election by splitting the African American vote or through voter intimidation tactics.<sup>139</sup> These boards made decisions on loan recipients, acreage allotments, appropriate crop yields, hardship adjustments, and preferred farming methods benefitting white farmers.<sup>140</sup>

The Farmers Home Administration was another agency that discriminated against and displaced African American farmers.<sup>141</sup> The agency offered loans and credit to poor farmers for home construction and improvement.<sup>142</sup> But instead of going to badly-needed rural housing in the South, these loans went to segregated swimming pools, picnic areas, tennis courts, and golf courses in white communities.<sup>143</sup> Loan requirements were stringent and often subjective, such as whether an applicant was a good citizen.<sup>144</sup> Loans went to the white and wealthy while African American farmers were turned down.<sup>145</sup> Even if an African American farmer received a loan, agency administrators would seek to get rid of them by luring them into debt and then foreclosing and auctioning off their machinery.<sup>146</sup>

As a result, African American farmers were pushed off their land.<sup>147</sup> Lawrence Lucas, President Emeritus of the United States Department of Agriculture Coalition of Minority Employees testified that the USDA's programs continue to discriminate against African American

### LAND OWNERSHIP AFRICAN AMERICAN FARMERS

In 1910 African American farmers owned **16 million acres** of land. In 2007, they owned **3.2 million acres**, an

# **80% loss**.

farmers and that "a culture of systemic racism at the USDA that denies Black farmers their dignity, that denies Black farmers a right to farm, denies Black farmers the right to the same programs and services that white farmers get in this country" is still present today."<sup>148</sup>

In 1910, Black farmers owned 16 million acres of land. In 2007, they owned 3.2 million acres, an 80% loss.<sup>149</sup> In 1999, Black farmers filed a class action lawsuit against the USDA for unlawful discrimination against them in denying their farm loan applications.<sup>150</sup> The lawsuit, *Pigford v. Glickman*, ultimately settled for money damages,

but no policy changes at the USDA.<sup>151</sup> While many claims have been paid, the USDA nonetheless has been slow to pay out all the claims, and has spent extensive resources in fighting claims.<sup>152</sup> In the 2021 coronavirus relief bill, \$4 billion was set aside for debt relief for socially disadvantaged

farmers, including African American farmers, but payments have been stopped do to an ongoing lawsuit alleging it is reverse-racism and a "windfall" for some farmers.<sup>153</sup>

## African American Wealth Destruction through Mass Incarceration

In the late 1980s mass incarceration and the war on drugs continued the American government's historical criminalization of African Americans.<sup>154</sup> As discussed in Chapter 11, An Unjust Legal System, African Americans have experienced marginalization, physical harm, and death, at the hands of the American criminal justice system at both the federal and state level beginning in slavery and continuing today.<sup>155</sup>

During the slavery era, federal and state governments criminalized African Americans as a method of establishing, maintaining, and socially controlling African Americans as a lower class of human being than white Americans.<sup>156</sup> Today, mass incarceration continue to separate families and dehumanize the descendants of enslaved African American people.<sup>157</sup> In the 156 years since slavery was abolished, African American people in the United States have gone from being considered less than human under the law to being treated as less than human by a criminal justice system that punishes them more harshly than white people.<sup>158</sup>

Until the 1940s, state laws and the U.S. Constitution allowed private entities to force African Americans into doing the same work, on the same land, and even for the same people as during slavery in a system called convict leasing.<sup>159</sup> People who were "leased" were treated even more brutally than enslaved people because plantation owners had a financial incentive to keep enslaved people alive.<sup>160</sup> Working and living conditions for incarcerated people were dangerous, unhealthy, and violent.<sup>161</sup> Most incarcerated people who were leased for labor did not survive to complete 10-year sentences.<sup>162</sup> Until the mid-1950s, states routinely forced chain gangs of imprisoned people to do public works projects while wearing chains weighing as much as 20 pounds.<sup>163</sup>

American politicians ran on "law and order" or "tough on crime" platforms and passed laws and policies

Mass incarceration creates a vicious cycle. Studies have shown that lower wealth increases the likelihood of incarceration and incarceration decreases the ability to build or maintain wealth.

> that punished African Americans more than white Americans, often for similar crimes.<sup>164</sup> This has contributed to mass incarceration and overrepresentation of African Americans in the criminal justice system to present day, nationwide and in California.<sup>165</sup> Nearly 70 percent of young Black men will be imprisoned at some point in their lives, and poor Black men with low levels of education make up a disproportionate share of incarcerated Americans.<sup>166</sup>

> Mass incarcerations create a vicious cycle. Studies have shown that lower wealth increases the likelihood of incarceration and incarceration decreases the ability to build or maintain wealth.<sup>167</sup>

> African Americans, who have lower family wealth than white Americans,<sup>168</sup> are especially vulnerable to incarceration.<sup>169</sup> Growing up with less family wealth means living in poorer neighborhoods with lower-quality

education and a greater exposure to high "street" crime and high imprisonment areas. Sixty-two percent of African Americans live in highly segregated, inner-city neighborhoods where socioeconomic vulnerabilities contribute to higher rates of violent crime,<sup>170</sup> while the majority of white Americans live in "highly advantaged" neighborhoods where there is little violent crime.<sup>171</sup>

Mass incarceration has been catastrophic to the ability of African American families to build and maintain wealth by reducing household assets and income, and lowering homeownership rates. As discussed in detail in Chapter 11, An Unjust Legal System, the criminalization of African Americans has contributed to racial disparities at every step of the criminal justice system.

In 2019, African Americans comprised 26 percent of all arrests yet they only made up 13.4 percent of the population.<sup>172</sup> According to a recent large-scale analysis of racial disparities in over 60 million state patrol police stops in 20 states, including California, researchers found that police officers stop African Americans more often than white drivers relative to their share of the driving-age population.<sup>173</sup> Further, these researchers found that, after controlling for age, gender, time, and location, police are more likely to ticket, search, and arrest African American drivers than white Americans.<sup>174</sup> Thus in practice, the bar for searching African Americans.<sup>175</sup>

Low family wealth can also mean being unable to afford additional education and delaying entering the labor market, leading to higher risks of incarceration.<sup>176</sup>

#### Once released, criminal convictions make it harder for returning citizens to find and maintain jobs, find leases, and be approved for mortgages.

Involvement in the criminal justice system increases legal debt.<sup>180</sup> Incarceration means loss of income and may lead to missed mortgage payments and other debts.<sup>181</sup> This increases interest obligations and penalties, which in turn can send an incarcerated individual's credit score plummeting.<sup>182</sup> Incarceration also means household instability, placing an additional barrier to building wealth.<sup>183</sup>

As Chapter 11 explains, federal and state prisons continue to exploit the labor of predominantly African American incarcerated people.<sup>184</sup> While convict leasing as an official practice has ended, underpaid or unpaid prison labor continues today as incarcerated people are not protected by labor laws.<sup>185</sup>

For example, in-house prison labor, the most common type of prison labor, typically refers to jobs within and related to the prison including kitchen duty, cleaning, or grounds keeping.<sup>186</sup> Workers can be punished and sent to solitary confinement for taking a sick day, including in the eight states where the labor is unpaid.<sup>187</sup> In states where prison labor is paid, the average rates across the U.S. range from 14 to 63 cents per hour. These are the rates before wage garnishment, which can account for up to half of one's earnings, although some advocates argue that wage garnishment serves as appropriate sources for victims' restitution.<sup>188</sup>

In addition to jobs within the prison, incarcerated people also provide underpaid and unprotected industry labor unrelated to maintaining the prison, like phone banking, packaging, and textile work.<sup>189</sup> Participating private companies must pay minimum wage, which

> the prison garnishes for the incarcerated person's room and board in prison cells.<sup>190</sup> The garnishment only applies to basic room and board, and the incarcerated person must pay additionally for stamps, paper, toiletries, supplementary food, or phone calls.<sup>191</sup> Advocates argue that such costs are not fairly

Once a person is a criminal defendant, low family wealth makes it hard to post bond or to hire lawyers to help navigate the criminal justice system, making incarceration more likely.<sup>177</sup>

In 2009, African Americans made up less than 13 percent of the U.S. population, but comprised over a third of all the people in prison.<sup>178</sup> As explained in Chapter 11, An Unjust Legal System, African Americans are also more likely to be convicted and experience lengthy prison sentences.<sup>179</sup> calculated and that the American criminal justice system is filled with fees that shift the costs of incarceration not only to the incarcerated, but also to their families.<sup>192</sup>

Scholars argue that work programs in incarceration are not beneficial to incarcerated people when they seek work after their incarceration.<sup>193</sup> Once released from incarceration, criminal convictions make it harder to find and maintain jobs, find leases, and be approved for mortgages.<sup>194</sup> A record of previous incarceration also has wide-ranging immediate and future consequences that act as a barrier to employment, thus lowering earnings.<sup>195</sup> A criminal conviction makes it difficult to build wealth because of stigmatization and lack of access to supportive social institutions and credit.<sup>196</sup>

## Exclusion and Discrimination in Government Benefits

African Americans have consistently been excluded from numerous major categories of government benefits, which have generally benefited white Americans. Government benefits refer to assistance programs

that provide either cash assistance or in-kind benefits to individuals and families from a governmental entity. There are two major types of government benefit programs: social welfare programs and social insurance programs. build wealth—insure their old age, get good jobs, acquire economic security, build assets, and gain middle-class status—most African Americans were left out.<sup>203</sup>

Social Security, passed and signed by President Roosevelt in 1935, left out most African Americans for the first quarter century of its existence.<sup>204</sup> Southern congressional representatives heavily influenced the Social Security Act with devastating results.<sup>205</sup> The federal government gave the power to control benefit levels, eligibility, and administration of such programs to the states and ex-

Historically, federal policy decisions dealing with welfare, work, and war excluded or discriminated against the vast majority of African Americans.

Benefits received from social wel-

fare programs are usually based on low income. Benefits received from social insurance programs are usually based on other criteria, such as age, employment status, or being a veteran. Some of the major federal, state, and local social insurance programs are Social Security, veteran's benefits, unemployment insurance compensation, and workers' compensation.

Social insurance programs can provide important support in times of crisis. Unemployment insurance, a state level program, helps protect against unexpected drops in income by paying cash benefits to unemployed workers who have lost jobs through no fault of their own.<sup>197</sup> The federal Supplemental Nutrition Assistance Program (SNAP), also known as food stamps, gives money to low-income families to buy food, and expands to provide important support when people lose their jobs.<sup>198</sup> Recipients have improved food security, health, and reduced healthcare expenses.<sup>199</sup>

Historically, federal policy decisions dealing with welfare, work, and war in the 1930s and 1940s excluded or discriminated against the vast majority of African Americans.<sup>200</sup> The exclusion and discrimination continued due to the Congressional representatives elected from southern states that created barriers to African Americans voting.<sup>201</sup>

Among other things, southern legislators prevented Congress from including anti-discrimination provisions in an expansive range of social welfare programs, such as community health services, school lunches, and hospital construction grants—all the programs that distributed funds to the South.<sup>202</sup> As a result, when federal policies provided most white Americans with valuable tools to cluded occupations mostly held by African Americans.<sup>206</sup> At the time, sixty-five percent of African Americans fell outside of eligibility for the Social Security program— and between 70 and 80 percent fell outside eligibility requirements in different parts of the South.<sup>207</sup>

For example, in 1940, the year Social Security payments for seniors began, the Social Security Board identified nearly 2.3 million African American workers as eligible for old age insurance, but the majority of those identified were disqualified because they were farm or domestic workers.<sup>208</sup> These exclusions continued until 1954, when the occupational exclusions were eliminated.<sup>209</sup>

This exclusion also left most African American workers out of unemployment insurance under the act.<sup>210</sup> When African American workers qualified by working in eligible industrial and commercial jobs, they often were left out because they lacked a history of regular, stable employment.<sup>211</sup> When they received benefits, the benefits tended to be smaller than those received by white workers.<sup>212</sup>

Another example of federal government discrimination in benefits are the mortgage subsidies, which not only intensified residential segregation as discussed in detail in Chapter 5, Housing Segregation, but also helped white families build wealth and enter the middle class. The federal government supported the creation and maintenance of the white middle class through other programs as well. The New Deal was a collection of government programs with the goal of lifting America out of the Great Depression.<sup>213</sup>

One example of a New Deal policy was the Serviceman's Readjustment Act of 1944, also known as the GI Bill,

which was reinforced in 1948 with the Integration of the Armed Forces Act. Through these laws, the federal government aimed to offer unprecedented benefits to veterans including mortgages to buy homes, job placement services, money for vocational and university education, and loans for small businesses.<sup>214</sup> However, these programs were administered

Social Security, passed and signed by President Roosevelt in 1935, left out most African Americans and scholars have argued that it was characterized "by a form of policy apartheid" for the first quarter century of its existence.

by the states, which discriminated against southern African American veterans.<sup>215</sup> While white World War II veterans sent themselves and their children to college and obtained housing and small business grants, African American veterans were not able to do so in the same way.<sup>216</sup>

Part of this stemmed from discrimination in the military. African American soldiers were more likely to be issued neutral and dishonorable discharges, sometimes used to exclude African American veterans from GI Bill benefits.<sup>217</sup> Ira Katznelson argues that severe discrimination in the GI Bill administration system prevented African American veterans from obtaining home mortgages, small business and farm loans.<sup>218</sup>

Chapter 10, Stolen Labor and Hindered Opportunity, discusses how other New Deal programs excluded African Americans in detail.

Today, African American families continue to have trouble accessing government benefits. Because welfare programs are often administered at the state and local levels, state and especially local governments have been able to introduce racial bias into welfare administration contributing to racially disparate outcomes.<sup>219</sup> States have been significantly more likely to both adopt and impose welfare sanctions if they have higher proportions of nonwhite welfare recipients.<sup>220</sup> States with higher African American populations—generally in the South—tend to provide fewer unemployment payments for a shorter time.<sup>221</sup> Additionally, in many places, part-time workers—who are disproportionately African American—are not eligible for unemployment payments.<sup>222</sup>

Despite having higher unemployment rates in general, African American workers receive unemployment benefits at lower rates than white Americans.<sup>223</sup> A report by the Government Accountability Office found that 73 percent of African American unemployment applicants received unemployment payments during the pandemic versus 80.2 percent of white applicants.<sup>224</sup> Although governments have waived work requirements for some SNAP recipients during certain national crises like the COVID-19 pandemic, studies have found that work requirements disproportionately cut off African American adults from SNAP benefits, which may be partially due to discrimination in the labor market making the job search more difficult for African American people.<sup>225</sup>



Protesters at the landmark civil rights demonstration, the March on Washington. (1963)

most households received several stimulus checks from the federal government. Studies have found that these payments were "likely crucial" to help households that lost jobs that pay their expenses.<sup>226</sup> The federal government supplemented state unemployment payments with up to \$600/week for unemployed workers, extended the duration of benefits, and gave benefits to workers traditionally left out of unemployment insurance programs.<sup>227</sup> The federal government also instituted the Paycheck Protection Program to provide loans that enable businesses suffering from COVID-19's economic shocks to pay their employees and other costs.228

During the COVID-19 pandemic,

During the pandemic, even though African Americans were more likely to hold the types of jobs most severely impacted by the pandemic, white households received their COVID-19 stimulus checks faster than African American households.<sup>229</sup>

This was likely due to the Internal Revenue Service focused structure of the program, which made receiving the payment more complicated for unbanked families and families who did not file taxes. African American people are more likely to be among both groups.<sup>230</sup>

Studies of pandemic-era federal loans have found that 29 percent of Black applicants were successful in obtaining loans for their businesses versus 60 percent of white applicants.<sup>231</sup> Businesses in majority-Black neighborhoods were also more likely to receive federal loans later than businesses in majority-white neighborhoods.<sup>232</sup> Federal

Black students carry more student loan debt because they receive a higher interest on their student loans and they borrow more because their families are less wealthy than white students.<sup>239</sup> For example, 20 to 29-year-old single white women who have completed college have a median net worth of \$3,400. Single African American women of a similar age and level of education have a median net worth of *negative* \$11,000.<sup>240</sup>

#### **Labor Market Discrimination**

Income is different from wealth. Income represents how much a person earns in a lifetime, both from work and from a yearly return on their investments.<sup>241</sup> Wealth represents a person's total net worth calculated from assets minus debts.<sup>242</sup> While income plays a modest role in the ability to generate wealth, as lower income translates to reduced capacity for savings or investments,

College degrees do little to close the racial wealth gap. For example, college-educated African American households have 30 to 33 percent less wealth at the median than non-college educated white households. Average wealth for white Americans in this category is \$180,500, while the African American average is \$23,400.

income does not explain massive Black-white wealth disparities in the United States. Without savings or wealth of some form, which can be passed from generation to generation, economic stability quickly falls apart when income is cut or disrupted through job loss, reduced work hours or reduced wages, or if families suffer from an unexpected health emergency.<sup>243</sup> In fact, the intergenerational transfer and impact

money was paid out through large banks, which historically excluded Black businesses.<sup>233</sup> African American business owners may have been less likely to obtain them, despite being more likely than white business owners to have at risk or distressed businesses even before the pandemic.<sup>234</sup>

#### **Education Segregation and Debt**

Higher education for African Americans can have a positive effect on their income, but does not translate into a reduction in the wealth gap.<sup>235</sup> College degrees do little to close the racial wealth gap.<sup>236</sup> For example, college-educated African American households have 30 to 33 percent less wealth at the median than non-college educated white households. Average wealth for white Americans in this category is \$180,500, while the African American average is \$23,400.<sup>237</sup> As Rucker C. Johnson testified before the Reparations Task Force, affluent African American households with incomes above \$75,000 still live in more under-resourced neighborhoods with under-resourced school than their white counterparts, which disadvantages African American students from the start of childhood.<sup>238</sup>

of wealth is one of the reasons why racial wealth inequities have become entrenched.<sup>244</sup>

As detailed in Chapter 10, Stolen Labor and Hindered Opportunity, labor market discrimination significantly contributes to the wealth gap.<sup>245</sup> Some scholars have based their estimates for reparations on more recent economic injustices such as labor market discrimination.<sup>246</sup> Bernadette Chachere and Gerald Udinsky estimated the monetary benefits that white workers gained from employment discrimination between 1929 and 1969.<sup>247</sup> They concluded that by the mid-1980s, white workers gained in \$1.6 trillion from employment discrimination at the expense of African American workers, assuming that 40 percent of the Black-white income gap was because of labor market discrimination.<sup>248</sup>

David Swinton concludes that even if one subtracted the total cost of government benefits programs including Social Security, Medicare, Medicaid, unemployment insurance, and other welfare programs—which are often argued to be reparations—over the same time span from the Chachere and Udinsky estimate, there would be still

be a \$500 billion net benefit to white people from labor market discrimination by the mid-1980s.<sup>249</sup>

But, income alone cannot explain the racial wealth gap. A reduction in racial differences in income would leave as much as three-fourths of the wealth gap unaddressed.<sup>250</sup>

Similar achievements do not lead to similar wealth for African Americans in comparison to white Americans.<sup>251</sup> For example, between 1984 and 2009, every dollar increase in average income for white households added \$5.19 in wealth. The same increase in average incomes for African American households added only \$0.69 in wealth.<sup>252</sup>

before the predatory subprime mortgage market crisis shows that there is no significant racial advantage in "passive savings" for white families with positive assets after family income is taken into account.<sup>261</sup>

#### **Discrimination in Banking**

African Americans have historically faced systemic discrimination in banking which has impacted their ability to accumulate wealth. Banks established by the federal government discriminated against African Americans and deprived them of wealth. The Freedmen's Fund, Free Labor and Union Army Military Banks, and the

Scholars estimate that 40 percent of the Black-white income gap is due to labor discrimination. As a result, by the mid-1980s, white workers gained in \$1.6 trillion at the expense of Black workers.

In fact, the racial wealth gap increases as income increases. The wealth gap between African Americans and whites in the bottom fifth of income levels is \$7,400, but the wealth gap between comparable African Americans and whites in the top fifth of income levels is \$264,700.<sup>253</sup> And, while white households have five to 10 times the net worth of African American households, they only earn twice as much as African American households.<sup>254</sup> Within the same income brackets, African American wealth is less than one-half that of white people.<sup>255</sup> White people in the bottom fifth of the income distribution have more than 10 times the median wealth of African American American American American American households.

Lower incomes for African Americans because of labor market discrimination affect wealth only to the extent that it reduces capacity for savings that can be passed across generations. There is no evidence that African Americans have a lower savings rate than white Americans once household income is taken into account.<sup>257</sup> One study found that once income is taken into account, if anything, African American families actually have a slightly higher savings rate than their white counterparts.<sup>258</sup> In fact, white households spend 1.3 times as much as African American households with similar incomes.<sup>259</sup>

At high income levels, African Americans save more than white people who tend to invest.<sup>260</sup> In addition to savings from income or "active savings," a family's wealth can also increase because of "passive savings" or when the value of a family's assets rises or appreciates. Data collected Freedman's Bank were three banking institutions established by the federal government in the early to mid-1860s, which provided recently emancipated African Americans with the means to save the money they earned.<sup>262</sup> But racist paternalistic attitudes by government officials prevented African

Americans from investing their own money and accumulating wealth.<sup>263</sup>Bank employees improperly invested client savings and lost approximately \$2.9 million, or \$63 million in 2017 dollars, harming freedmen and their descendants for generations to come.<sup>264</sup>

In another example, Union army chaplain John Eaton created the Freedmen's Fund in 1862, to hold the wages of formerly enslaved African Americans who fled to Union or who worked for Union troops.<sup>265</sup> These freedmen had no access to their individual wages or savings, nor did they have any say in how their own wages or the money that was donated for their benefit would be used.<sup>266</sup> Instead, Eaton pooled the wages in the fund to collectively provide food, shelter, and other needs, essentially treating freedmen as contract laborers.<sup>267</sup>

Eaton also stole all their wealth. Soldiers confiscated horses, wagons, money, and other valuables that self-emancipated African Americans brought with them to the Union lines.<sup>268</sup> Eaton took anything that the soldiers and quartermasters did not steal for themselves.<sup>269</sup> By 1864, he had formalized his contract labor system to negotiate contracts for and hired out Black workers on abandoned plantations that the federal government had leased to northerners and to some southerners who supported the Union.<sup>270</sup> Any profits from the cash crops that Black workers grew and harvested were placed in the Freedmen's Fund.<sup>271</sup> Eaton also used the fund for other expenses.<sup>272</sup> In one year alone, Eaton stole \$103,000 or \$1.6 million in 2017 dollars from Black depositors to pay for the Union Army's incidental expenses; \$5,000 in medical expenses; the salaries of all hospital stewards and medical assistants. <sup>273</sup> The Free Labor and Union Army Military Banks first established in 1864, was another Bank that used an exploitative model of contracting African American people's labor similar to Eaton's freedmen's fund.

In 1865, Congress created the Freedman's Bank and Trust Company, also known as the Freedman's Savings Bank, seeding the bank with unclaimed deposits from the free labor and military banks.<sup>274</sup> The initial charter designated an all-white board of trustees with broad discretion to oversee the bank, and intended to hold only the deposits of the survivors of enslavement and their descendants.<sup>275</sup> Despite this nominal limitation, the bank welcomed customers of all races and regardless of whether they were formerly enslaved, though formerly enslaved people made up the vast majority of bank customers.<sup>276</sup> And though the charter made clear that its purpose was to invest the deposits in low-risk treasury notes and conservative U.S. securities, it vaguely stated that a third of the deposits, called "available funds," could be invested anywhere, leaving an opening for abuse.277

The Freedman's Bank used a number of aggressive methods and tactics to solicit deposits and to convince African American patrons that their money was safe and that they could grow wealth.<sup>278</sup> Passbooks and other bank literature contained numerous slogans and poems on the ways of thrift and savings.<sup>279</sup> Bank advertisements often included the names of prominent government

officials, such as Abraham Lincoln and Oliver Otis Howard, misleading customers and the public into believing that the federal government protected and guaranteed their deposits.<sup>280</sup> Depositors were reminded during public meetings and other bank-sponsored gatherings that the bank was under Congressional charter, and thus under its complete protection.<sup>281</sup>

With such assurances that their deposits were safe, African Americans from a wide variety of backgrounds and occupations, many excited to be receiving a wage for their services for the first time in their lives, opened accounts with the Freedman's Bank between 1868 and 1874 at an extraordinary rate. Within 10 years, 75,000 depositors—who were virtually all African Americans trusted the bank by depositing more than \$75 million, approximately \$1.5 billion in today's dollars.<sup>282</sup> Most of these deposits were being saved to buy land and other productive goods such as tools or agricultural supplies as depositors were told to do.<sup>283</sup>

But the bank turned quickly from a savings bank to a risky private investment bank controlled by a small minority of trustees.<sup>284</sup> Against the bank's original Congressional charter and without the knowledge of the African American customers, who were largely unable to secure loans from the bank, the bank invested the money in risky railroad and real estate holdings to benefit white businessmen and bank managers.<sup>285</sup> When, on June 29, 1874, the bank failed and closed due to fraud, 61,131 mostly African American depositors lost about \$2.9 million or \$63 million in 2017 dollars..<sup>286</sup> One study estimated the average amount owed to depositors across 71 bank failures of federally chartered banks between 1865 to 1933,<sup>287</sup> and the Freedman's Savings Bank ranked third for the largest amount owed to depositors at the time of bank failure.<sup>288</sup>

Because the bank had represented much more than just a place to store money, the African Americans who lost their money also lost their trust in the federal government and in banks in general.<sup>289</sup>

State and private banks following emancipation refused to serve the credit needs of freedmen during the late 19th century, which meant that they had to rely on more expensive and exploitive credit systems.<sup>290</sup> General stores became an important means of accessing short-term credit.<sup>291</sup> Prices were at the discretion of the merchant.<sup>292</sup> One price for goods purchased with cash

Bank officials at the government charterd Freedmen's Bank improperly invested the savings of newly freed enslaved people, leading to the loss of approximately \$2.9 million or \$63 million in 2017 dollars. This loss of wealth devastated newly freed enslaved people and their descendants for generations.

> and a higher price (often by 25 percent) for goods purchased with credit.<sup>293</sup> Goods purchased on credit were charged interest of eight to 15 percent, as determined by the personal judgment of the merchant, based on the creditworthiness of the borrower.<sup>294</sup>

> Black-owned banks were established to provide banking services to Black communities.<sup>295</sup> Approximately 130 Black-owned banks were established between 1900 and 1934. Fifty savings and loans and credit unions were also established during this period. Only eight banks

survived the Great Depression out of 130 Black-owned banks.<sup>296</sup> Today, there are only 21 Black-owned banks nationwide, and 32 Black-owned financial institutions overall, including credit unions.<sup>297</sup>

The federal government prevented the success of Blackowned banks by excluding them from full participation in the banking market.<sup>298</sup> African American bank deposits were smaller and were more frequently withdrawn than white bank deposits, which made them more risky.<sup>299</sup> Most African American depositors had no wealth to invest in the bank and were just depositing money from their wages while keeping small amounts to live on.<sup>300</sup> They put their money into Black-owned banks not only for safe-

keeping, but also as rainy-day funds during bad times that came often.<sup>301</sup>

needed for safe, and profitable banking.<sup>307</sup> Most thriving banks hold a mix of commercial and real estate loans, but Black-owned banks made almost exclusively home loans because the vast majority of African American businesses were small service operations with no need for bank financing.<sup>308</sup>

As described in Chapter 5, Housing Segregation, the federal government generally labeled African American homeowners and African American neighborhoods as being at higher risk of default, and white-owned banks generally refused to issue mortgages to African American homeowners. Black-owned banks often met the need and provided home loans to African American homebuyers.<sup>309</sup>

Because their deposits experienced high risk, Black-owned banks had to keep more cash as reserves or invest in other more liquid assets such as

government securities, which were safer than loans.<sup>302</sup> They needed to make sure they always had enough cash at the bank to pay out to depositors. Black-owned banks also held very high capital ratios to offset this risk.<sup>303</sup> For example, in 1920, the mean capital ratio for white banks was 18 percent, while African American banks had an average capital ratio of 32.9 percent.<sup>304</sup> This meant that the African American bank owners invested more of their own money and earnings in the bank to keep it secure, but this severely lessened their ability to make a profit or lend money.<sup>305</sup>

Another source of vulnerability for African American banks was their assets or loan portfolios.<sup>306</sup> The fate of African American banks was tied up with the fate of African American businesses, which meant that African American banks lacked the diversified investments

COURTESY OF JASON ARMOND / LOS ANGELES TIMES VIA GETTY IMAGES



Incarcerated firefighters from Eel River Conservation Camp continue to tackle the Caldor Fire as the fire's footprint continues to expand southwest of the Lake Tahoe Basin. (Aug. 27, 2021)

Before the Great Depression, there were 130 Black-owned Banks. Today there are only 21 Black-owned banks.

> Since homes owned by African Americans were undervalued due to government redlining, the property held for collateral during the term of the loan immediately diminished in value, upholding the perception that these loans were inherently risky investments.<sup>310</sup> Therefore, there was no market for mortgages held by African Americans because of the devaluation of property owned by African Americans and the assumption that loans held by African Americans were inherently risky.<sup>311</sup> This in turn meant that it was difficult for Black-owned banks to earn a profit from an investment portfolio that was largely composed of home loans to African American homebuyers.<sup>312</sup>

#### California

In California, Black homesteaders can be traced back to 1900, when agricultural settlements were promoted at various times after the turn of the century in Yolo, San Bernardino, Tulare, and Fresno counties.<sup>313</sup> At least two different efforts at colonization occurred in San Bernardino County between 1900 and 1910, including solicitation of families to homestead government land in the Sidewinder Valley, desert land near Victorville.<sup>314</sup> Black homesteaders also established an agricultural settlement in 1908 in the town of Allensworth in Tulare County, which ultimately was depleted of a water supply necessary to sustain the settlement.<sup>315</sup>

The racial climate around African American colonies ranged from welcoming or neutral to hostile, although none have been reported to experience the kind of everyday violence and intimidation African Americans regularly experienced in the South.<sup>316</sup> Incarcerated people in California produces myriad products such as clothing, furniture, cleaning products and food.<sup>317</sup> They also perform a wide range of duties in areas such as laundry, kitchen, and general maintenance.<sup>318</sup> The California Department of Forestry and Fire Protection employed around 1,600 incarcerated individuals to fight forest fires in May 2021.<sup>319</sup> Some are paid as little as \$1.45 a day.<sup>320</sup> As discussed in Chapter 10, An Unjust Legal System, advocates argue that this is exploitive and does not necessarily help the incarcerated firefighters to be find jobs once they are released.<sup>321</sup>

The costs of higher education are a larger burden for African American Californians.<sup>322</sup> Generally, white Americans are twice as likely as African Americans to receive financial assistance from their families for higher education.<sup>323</sup> Only 16 percent of very low-income African American Californian students receive a CalGrant award.<sup>324</sup> The state financial aid African Americans do receive is often insufficient, especially with respect to housing.<sup>325</sup> Fifteen percent of white households in Los Angeles had student loan debt, in contrast with 20.5 percent of households headed by African Americans.<sup>326</sup>

Structural racial disparities regarding access to unemployment insurance, food stamps, and COVID-19 federal loans in a crisis, and benefits for businesses also exist in California. From March through June 2020, 84.9 percent of California's African American labor force filed for unemployment benefits, compared to 39.1 percent of the state's white labor force.<sup>327</sup> African American Californians who received unemployment insurance during the pandemic received \$293.90, the smallest median weekly benefit of any racial group, versus white claimants who received \$394.90.<sup>328</sup>

Generally, white students are twice as likely as Black students to receive financial help from their family for higher education. Only 16 percent of very low-income Black Californian students receive a CalGrant award.

California's version of food stamps, CalFresh, generally maintains the same work requirements that disproportionately cut African American adults off from food assistance at the federal level.<sup>329</sup> In 2016, California enrolled only 72 percent of eligible residents in CalFresh, the fifth lowest rate in the nation.<sup>330</sup> It is also one of 10 states that manage food assistance programs at the county level, which tends to be more expensive and variable than administering the program at the state level.<sup>331</sup> African American Californians make up 6.5 percent of

the state's population,<sup>332</sup> but 14.7 percent of participating CalFresh households.<sup>333</sup> During the pandemic, one survey found that 20.2 percent of Black households with children sometimes or often did not have enough to eat in a four week period spanning June and July, compared to 8.8 percent of white households with children.<sup>334</sup>

#### CALIFORNIA UNEMPLOYMENT INSURANCE MARCH TO JUNE 2020

Weekly benefit amount

White	\$394.90
African American	\$293.90

In California, an analysis of the distribution of federal loans found disparate distribution by race: African American neighborhoods received \$445 per resident, while white neighborhoods received \$666 per resident, partially due to lower concentration of small businesses or small business employees in African American neighborhoods.<sup>335</sup> However, another analysis revealed that in most major metro areas in the country—including Los Angeles, San Francisco, and San Diego—businesses in majority-white areas also received federal loans at a greater rate than businesses in majority-African American areas.<sup>336</sup>

Proposition 209 (Prop 209) has also disadvantaged Black-owned businesses in the state seeking public contracts with the State of California and local governments. Passed in November 1996, Proposition 209 caused state and local governments to end race-con-

> scious contracting programs, resulting in a loss of \$1 billion to \$1.1 billion every year for minority and women-owned businesses.<sup>337</sup> The biggest contract loss for minority and women-owned businesses was with the State of California where \$823 million has been lost each year since Prop 209.<sup>338</sup> Before Prop 209 passed, in fiscal year 1994-1995,

\$519 million was contracted to minority and women-owned businesses or about \$823 million in October 2014 dollars. When California ended its program for minority and women-owned businesses, only a few got back their contracts with the state and some never recovered.<sup>339</sup> There was only an insignificant increase in Small Business Enterprise procurement with the state after Prop 209, which is the main way that state contracts would be available for many of these businesses.<sup>340</sup> The City and County of San Francisco lost about \$200 million per year in minority and women-owned business contracts. <sup>341</sup> Some of this loss emerged immediately after Prop 209. More losses followed the 2004 Coral Construction case, which ultimately ended San Francisco's race-conscious procurement program. Prop 209 also resulted in the loss of about \$30 million per year in minority and women-owned business contracts with the City of Oakland. It also resulted in an estimated \$20 million loss per year in such contracts with the City of San Jose after the 2000 Hi-Voltage Wire Works case, which ultimately ended San Jose's race-conscious contracting program.<sup>342</sup>

## IV. Drivers of the Wealth Gap Today

#### **Unequal Homeownership**

Homes are one of the most important wealth assets that households can possess.<sup>343</sup> People who own homes can use them to borrow money to pay for expenses or pay off high-interest debt in times of crisis.<sup>344</sup> Homeowners are able to generate wealth through home equity, so a crisis. The racial homeownership gap was 19 percent in 1940, and grew to 28 percent in 2009. As of the second quarter of 2020, out of \$30.8 trillion in real estate assets in the U.S., Black households held five percent and white households held 78 percent.<sup>353</sup> In 2019, 42.8 percent of African Americans owned homes versus 73.3

Nationally, In 2019, 42.8 percent of African Americans owned homes versus 73.3 percent of white Americans. Only 33 percent of Black Californians owned homes. percent of white Americans, and are more likely to face affordability issues in securing capital to purchase and sustain housing at 30 percent of their gross income, including utilities.<sup>354</sup> In the same year, researchers for the National Bureau of Economic Research also found that African American mortgage bor-

long as their home increases or appreciates in value. Homeownership is also believed to be more beneficial than renting because owners build equity, and obtain additional tax benefits.<sup>345</sup> Homeowners may also face less housing instability than renters—partially because they tend to be more well-off in general—especially during a crisis, and may therefore be less likely to lose their housing.<sup>346</sup> Housing affordability problems where an occupant must pay more than 30 percent of gross income for housing costs, including utilities<sup>347</sup>—are more than twice as common among renters than homeowners.<sup>348</sup>

As discussed in Chapter 5, Housing Segregation, and above, government discrimination made it difficult for African Americans to buy real estate, gain wealth through real estate, and transfer that wealth to successive generations.<sup>349</sup> Widespread homeownership in the United States was created through government action, starting with New Deal legislation.<sup>350</sup> The New Deal created relatively safe long-term mortgage markets and reduced down payment requirements for homeownership. This transformed the housing landscape, allowing many working-class households to move from the rental lifestyle to owning a home.<sup>351</sup> Yet, as described above and in Chapter 5, Housing Segregation, the path to homeownership has been riddled with entrenched racism.<sup>352</sup>

Today, African Americans are in a worse position than white Americans to have homes as assets to aid them in rowers were charged higher interest rates than white borrowers were and were denied mortgages that would have been approved for white applicants.<sup>355</sup>

African Americans who own homes have a greater reliance on the house as a source of wealth than white households.<sup>356</sup> In 2014, home equity accounted for 92 percent of African Americans' net worth.<sup>357</sup> There is a gap between the appreciation of a home owned by a white family and the appreciation of a similar home owned by an African American family.<sup>358</sup> When African Americans do own homes, they tend to be appraised for less than comparable white homes, limiting the amount of money that can be taken out of their home equity. Race affects the rate of return on home asset.<sup>359</sup> African American



homeowners had a median home equity of \$66,800 in 2019.  $^{360}$  White homeowners had a median home equity of \$130,000 in the same year.  $^{361}$ 

Residential segregation contributes to the undervaluing of houses in African American neighborhoods.<sup>362</sup> African American homeowners tend to own homes appraised for less in neighborhoods deemed less valuable,<sup>363</sup> which decreases their available equity relative to white homeowners.<sup>364</sup> Even controlling for factors like neighborhood or home quality, a study has found systemic undervaluation of homes in African American neighborhoods attributable to anti-Black bias.<sup>365</sup> Major companies offering real estate insurance have been accused of targeting inner city neighborhoods where Black families live by denying claims as fraudulent.<sup>366</sup> All of this limits African Americans' access to the benefits of home equity in a crisis.<sup>367</sup>

African Americans also experience significant housing cost burdens.<sup>368</sup> Without sufficient wealth in the first place, African American households have limited means to invest in homeownership.<sup>369</sup>

In 2019, 43 percent of African American households spent more than 30 percent of their income on housing—compared with 25 percent of white households.<sup>370</sup> Discrimination in mortgage lending may also make it more difficult for Black homeowners to access their home equity through cash-out refinancing,<sup>371</sup> a means of accessing home equity that has been increasingly popular during the pandemic.<sup>372</sup> Between April 2020 and January 2021, less than a quarter of African American homeowners who could have saved \$200 a month by refinancing did so, compared to 40 percent of similarly situated white homeowners.<sup>373</sup>

But closing the homeownership gap alone will not close the racial wealth gap; the homeownership gap alone does not explain the racial wealth gap.<sup>374</sup> Among African American and white American households who do not own a home, white households still have 31 times more wealth than African American households.<sup>375</sup>

#### **Fewer Assets**

African American households hold less assets than white households overall, but African American households hold a higher proportion of assets in their cars and homes, and less in net liquid and net business assets.<sup>376</sup> African American households are also generally less likely to hold financial assets.<sup>377</sup> African Americans have substantially fewer assets than white people at every income level, including bank deposits, stocks, bonds, and loans.<sup>378</sup> African Americans tend to have fewer investments. Some studies argue that African American investment patterns generally show risk aversion and lack of education on stocks and investments.<sup>379</sup> They argue that wealthier African Americans tend to save more and invest less<sup>380</sup> compared to wealthier white Americans, and that white Americans are more likely than African Americans to invest in high-risk, high-reward assets.<sup>381</sup> For example, in 2004, African American families were less likely than white families to have investment accounts and retirement accounts. Only 44 percent of African Americans have retirement savings accounts, with a typical balance of around \$20,000, compared to 65 percent of white Americans, who have an average balance of \$50,000, according to the Federal Reserve.<sup>382</sup> And only 34 percent of African Americans own any stocks or mutual funds, compared to more than half of white people.<sup>383</sup> Some studies claim that this can be attributed to familial influence-Black families are less likely to have investment accounts if their parents didn't have any.<sup>384</sup> Other studies argue that African Americans are not significantly more risk averse or less financially literate than white people with similar levels of income and wealth. Further, African Americans engage in entrepreneurship, which presents inherent risk, at higher rates than white Americans with similar levels of income and wealth. Low wealth, lack of financial inclusion, and financial constraints on choice often forces Black borrowers to use predatory and abusive alternative financial services rather than financial illiteracy.385

#### 2019 LIQUID ASSETS AMOUNT BY RACE



African Americans have less liquid assets. Liquid assets accessible as cash in times of crisis include cash savings, checking accounts, savings accounts, money market funds, certificates of deposit, and government bonds.<sup>386</sup> Access to liquid assets is important in a crisis, as it enables people to continue to pay bills in the event of a sudden loss of income, or pay for emergency expenses such as medical costs. Lack of access to liquid assets can

heighten the impact of crises by making it harder to afford basic necessities.<sup>387</sup> People may also turn to family for economic support in times of hardship.<sup>388</sup> In addition, access to government aid such as unemployment insurance, nutrition subsidies, and crisis-specific programs, such as stimulus checks and small business loans, help people and their businesses stay afloat.<sup>389</sup> These resources are vital for surviving economic crises. For example, liquid assets such as cash savings help people pay bills in the event of a job loss or weather emergency expenses like a medical emergency.<sup>390</sup> Similarly, people who have homes, stocks, or retirement funds may leverage their home value for a loan,<sup>391</sup> liquidate stocks, or borrow from or against their retirement accounts to pay for expenses during difficult economic times.<sup>392</sup>

African American households tend to disproportionately lack access to many of these resources, often due

to the persistence of historical disparities and racism that continues today.<sup>393</sup> In 2019, while 96.8 percent of African American familieshadsomekindofliquidasset such as a checking account, savings account, or pre-paid card—typical African American families with liquid assets had \$1,500 in liquid savings, compared to \$8,100 for white families with liquid assets.<sup>394</sup> Racism in the banking system today still create barriers to liquid assets for African American customers. African emergency, compared to \$14,400 for the typical African American family.  $^{403}$ 

While African Americans are more likely to have access to retirement accounts than homes or other types of stocks, they are still less likely than white Americans to have them.<sup>404</sup> Around 55 percent of African American working-age families have access to an employer-sponsored retirement plan, and 45 percent participate.<sup>405</sup> Seventy percent of white working-age families have access to an employer-sponsored retirement plan, and 60 percent of them participate.<sup>406</sup> Among working-age white families with balances in such accounts, the typical white family has approximately \$50,000 saved, whereas the comparable African American family has approximately \$20,000 saved.<sup>407</sup> During the pandemic, a survey found that African American households with retirement accounts were much more likely to

African American business owners typically start their businesses with half the capital of white business owners despite the fact that they demonstrate a greater need for start-up financing. The median loan amount for African American business owners who are approved for credit is less than half of the loan amounts extended to their white counterparts.

American customers are sometimes profiled,<sup>395</sup> viewed with suspicion just for entering a bank,<sup>396</sup> and questioned<sup>397</sup> over the most basic transactions.<sup>398</sup>

African Americans have less non-liquid assets. In general, non-liquid assets such as homes, stocks, and retirement funds can support financial security by increasing resources necessary to weather a crisis or invest in wealth-generating assets for the future.<sup>399</sup> As discussed above, African Americans experience myriad barriers to homeownership and the mortgage market. Stocks and mutual funds, which can be sold, and retirement funds, which can be liquidated or borrowed against, also provide potential sources of aid in a crisis.<sup>400</sup> African Americans are also less likely to own stocks than white Americans, and African Americans who own stocks have less equity than white Americans do.<sup>401</sup> While 61 percent of white households own any form of stocks, only 33.5 percent of African American households do.<sup>402</sup> Among families who own stocks, the typical white family has access to \$50,600 they could tap in an report that they planned on withdrawing from or borrowing against them (48 percent and 45 percent) than white households (29 percent and 29 percent) due to relative lack of other assets.<sup>408</sup> However, withdrawing money from retirement accounts can incur tax and other penalties.<sup>409</sup>

#### **Lower Business Ownership**

Business ownership allows African Americans to participate in local, regional, and global markets from which they have historically been excluded due to systemic racism and discrimination.<sup>410</sup> Equity in a business is among one of the types of assets that are more unequally distributed by race.<sup>411</sup> Lower wealth for African Americans leads to lower business ownership and self-employment.<sup>412</sup> Studies have demonstrated the substantial wealth advantages to self-employment and have shown that those who become self-employed show much stronger gains in wealth compared to individuals who never become self-employed.<sup>413</sup>

#### **UNITED STATES** 2004 BUSINESS OWNERSHIP BY RACE



This is especially true for Black business owners given that the median net worth for Black business owners is 12 times higher than Black non-business owners.<sup>414</sup> Further, Black business ownership is a viable path to creating wealth not only for Black business owners, but also for Black communities at large.<sup>415</sup> Most small businesses tend to hire from the community, which tends to create job opportunities for community residents.<sup>416</sup> Therefore, the success of Black-owned businesses is a critical path for economic empowerment in Black communities.<sup>417</sup>

The Center for Financial Household Stability and the Federal Reserve Board of St. Louis have documented that, compared to white individuals, African Americans own fewer of their assets in the form of business assets.<sup>418</sup> In terms of market share, Black-owned businesses are significantly underrepresented in comparison to white and other minority-owned businesses.<sup>419</sup> In 2017, only 3.5 percent of all United States businesses were Black-owned compared to 81 percent white-owned.<sup>420</sup> Although Black business ownership has been steadily increasing in the United States, growth has been tremendously slow.<sup>421</sup> Several factors contribute to the racial disparity in American business ownership such as systemic barriers to securing start-up capital and the relatively small size of Black businesses.<sup>422</sup>

sought financing for planned business growth or expansions compared to only 55 percent of all respondents.<sup>425</sup> Although African American businesses demonstrate a greater desire and need to secure financing, due to discriminatory lending practices, African American business owners receive lower loan amounts, and with less frequency than white business owners.<sup>426</sup> For example, the median loan amount for African American business owners who are approved for credit is less than half of the loan amounts extended to their white counterparts.<sup>427</sup>

Another barrier to the growth and development of African American businesses is the fact that, on average, the businesses owned by African Americans are smaller than those owned by white Americans.<sup>428</sup> A key factor for measuring the size of a business is whether the business has employees and statistics show that Black-owned businesses are much less likely to have employees than white-owned businesses.429 In 2012, for example, 23.9 percent of businesses owned by white men had employees whereas only six percent of businesses owned by African American men had employees.<sup>430</sup> Although the size of a business has a significant influence on the profitability of a business, even Black-owned businesses with employees tend to be much less profitable than white-owned businesses with employees.<sup>431</sup> In 2014, 63.4 percent of white-owned businesses with employees indicated that they were profitable compared to the reported profitability of only 45.6 percent of Black-owned businesses with employees.432 In addition, the top 100 Black-owned businesses combined earned less than \$30 billion in 2014. Walmart earned \$482 billion, or sixteen times that.433

The challenges faced by African American business owners were further exacerbated by the economic hardships

COURTESY OF ARCHIVE PHOTOS VIA GETTY IMAGES

African Americans face many systemic barriers when seeking the social and financial capital necessary to start their own businesses that make it increasingly difficult for African American entrepreneurs to secure the financial capital necessary to launch or grow their own businesses.423 As a result, African American business owners typically start their businesses with half the capital of white business owners despite the fact that they demonstrate a greater need for start-up financing.424 According to Pepperdine's Private Capital Access Index report, approximately 80 percent of African American businesses



First Black Owned Aamco Transmission Franchise (c. 1970)

caused by the COVID-19 pandemic.<sup>434</sup> The Federal Reserve Bank of New York reports that about 58 percent of Blackowned businesses were at risk of financial distress before the pandemic, compared to approximately 27 percent of white-owned businesses.<sup>435</sup> The financial instability experienced by Black-owned businesses made these businesses particularly vulnerable at the onset of the pandemic.<sup>436</sup> According to a report by the House Committee on Small Business, Black business ownership declined approximately 40 percent, more than any other racial group, during the first few months of the pandemic.<sup>437</sup> Black business owners such as Richard Anderson, the owner of Kinfolk Brass for white Californians.<sup>442</sup> African American Californians less likely to have access to the credit and housing stability that owning a home can provide in a crisis.<sup>443</sup> Homeownership for African American Californians lags behind the nationwide African American homeownership rate–33.3 percent versus 44 percent nationally in 2019.<sup>444</sup>

63 percent of white Californians own their homes, while only 33 percent of African American Californians do.<sup>445</sup> Homeownership rates for African American households have fallen every decade for the last 30 years, both un-

> conditionally and after controlling for income and demographics.<sup>446</sup> African American Californians face

> higher priced loans, more pred-

atory lending, and more risk.<sup>447</sup> During the housing crisis of the

2000s, California had the country's

highest foreclosure rates, with Los Angeles leading the state.<sup>448</sup> African

During the housing crisis of the 2000s, California had the country's highest foreclosure rates, with Los Angeles leading the state. African American household foreclosure rates were 1.9 times that of white Americans, due to discriminatory banking practices.

Band and Music Group, experienced significant economic distress as a result of the pandemic.<sup>438</sup> Before the pandemic, Kinfolk Brass Band was one of the most popular bands in New Orleans and frequently performed at weddings and music festivals around the world.<sup>439</sup> However, lockdown measures enacted to reduce the spread of COVID-19 forced all large events and gatherings to cease.<sup>440</sup> Without any events to perform at, Kinfolk Brass Band and its band members, including owner Richard Anderson, suffered a significant loss of income in 2020.<sup>441</sup>

#### California

African American Californians are much more likely to be renters and much more likely to be housing cost-burdened. For instance, 58.4 percent of African American Californians are renters versus 34.1 percent American household foreclosure rates were 1.9 times that of white Americans, likely due to increased targeting of minority communities for predatory lending as discussed above.<sup>449</sup>

African American Californians have less non-liquid assets. African Americans in Los Angeles own fewer stocks, mutual funds, investment trusts, individual retirement account and/or private annuity than white Angelinos.<sup>450</sup> Eighteen percent of U.S.-born Black households in Los Angeles do not have a car, the lowest of all reported ethnicities.<sup>451</sup> While Californians, in general, face long commute times, African American Californians in Los Angeles, on average, have a 7.5 percent longer commute.<sup>452</sup>

In Los Angeles, 11.7 percent of white households own a business versus 3.1 percent of African American households.<sup>453</sup>

## V. Effects of Wealth Gap

The harmful effects of the wealth gap, which cascade across generations, have resulted in racial differences in the capacity of African Americans to transmit resources across generations and lower financial resilience during crisis, and discriminatory tax structures.

#### Fewer Intergenerational Wealth Transfers

Lower assets of African Americans means that intergenerational wealth transfers are less likely and tend to be smaller. Inheritance, intergenerational wealth transfers, or parental wealth are primary sources of the capacity for sustained wealth building.<sup>454</sup> Wealth, more than income, can be used to invest in appreciating assets for children, such as a college education, an unpaid internship in a high rent city, a new business, a property in a better residential neighborhood, or a job in the family firm.<sup>455</sup> Without wealth transfers, regardless of income, these assets are harder to attain.<sup>456</sup>

The fewer resources the older generation has to transfer to the next, the lower the wealth position attained by the younger generation.<sup>457</sup> At least 26 percent of an adult's wealth position is directly due to inheritance or gift money—a conservative estimate. The true effect could be as high as 50 percent.<sup>458</sup> Greater familial assistance contributes to parents have more resources, for example, to help them with down payments on their first house or to help them pay off their student loans.<sup>467</sup> About 80 percent of African American millennials with at least a bachelor's degree

white families' greater ability to buy better housing and get better deals on mortgages earlier in life, further compounding the homeownership and wealth gap and giving white families better security in crisis.<sup>459</sup> An Urban Institute study estimates that the shortfall in large gifts and inheritances accounts for 12 percent of the Black-white wealth gap.<sup>460</sup>

Between 26-50 percent of an adult's wealth position is directly due to inheritance or gift money. Larger family contributions earlier in life help white families buy better houses and get better mortgages, and create better financial security to combat a crisis.

The impact of fewer intergenerational transfers is reflected in the wealth gap between African American and white American millennials. While the typical white millennial family has about \$88,000 in wealth, the typical African American millennial family has only about \$5,000 in wealth. White millennial families made huge strides between 2016 and 2019, and they now lag previous generations of white families by only about five percent.<sup>461</sup> Between 2007 and 2019, however, African American millennials fell further and further behindnot just compared with white millennials, but compared with previous generations of African Americans.<sup>462</sup> While white millennials trail the wealth of previous generations of white Americans by only five percent, African American millennials trail previous generations of African Americans by 52 percent.<sup>463</sup> The typical African American millennial has \$5,700 less in net worth than counterparts in previous generations.<sup>464</sup>

There are several reasons for these deep disparities between African American and white millennials.<sup>465</sup> First,

In 2019, 71.9 percent of white families expected that they could get \$3,000 from friends or family during a crisis, versus less than 40.9 percent of African American families. Lack of access to liquid assets can also force people into financially risky options during a crisis, such as taking out predatory payday loans or high-interest credit card debt.

white millennials are more likely to benefit from having wealthy parents. Greater familial assistance contributes to white families' greater ability to buy better housing and get better deals on mortgages earlier in life, further compounding the homeownership and wealth gap and giving white families better security in crisis.<sup>466</sup> Their still have student loan debt, compared with about half of white millennials. White millennials are also more likely to own assets like stocks and homes, which have ballooned in value in recent years.<sup>468</sup> While about twothirds of white millennials own homes, less than a third of African American millennials own homes.<sup>469</sup>

#### **Lower Financial Resilience during Crisis**

Support from family networks can provide a "private safety net" to aid with cash transfers, housing, or child-care in times of material hardship.<sup>470</sup> Cash transfers can provide additional income, multigenerational housing can provide shelter, and family-provided childcare can permit a parent to work and earn income as well as avoid childcare expenses.<sup>471</sup> For example, during the COVID-19 pandemic, almost a quarter of renters borrowed money from friends or family.<sup>472</sup>

While African Americans receive assistance from family members at high rates, their overall tendency to lack resources may reduce the available quantity of such

> assistance, and may result in economic harm to the giver.<sup>473</sup> African American families are more likely than white families to have high-poverty family networks and more likely to make repeated cash transfers, which hinders their ability to accumulate wealth.<sup>474</sup>

> For example, in 2019, 71.9 percent of white families expected that they could get \$3,000 from friends or family during a crisis, versus

less than 40.9 percent of African American families.<sup>475</sup> While new African American mothers are more likely to live in a relative's home, host a family member, and give or receive money than white mothers, helping family members in poverty may have negative consequences for struggling families.<sup>476</sup>

A pre-pandemic study found that while 31.7 percent of white households are liquid-asset poor (meaning that they could not use their savings to live for three months at the federal poverty rate), 62.7 percent of African American households are.<sup>477</sup> And a 2020 study found that 36 percent of white families had enough savings to cover six months of expenses, versus 14 percent of African American families.<sup>478</sup> In one February 2021 survey of "disadvantaged workers," 42 percent of white households reported that they could not pay for a \$400 emergency expense without taking on additional debt, drawing down retirement accounts, or selling items, compared to 59 percent of African American households.<sup>479</sup>

African Americans are more likely to suffer from economic crises such as the COVID-19 pandemic. For example, a June 2020 survey found that while only 27 percent of white households had experienced financial hardship as a result of the pandemic, 40 percent of African American households had.480 In addition, lack of access to liquid assets can also force people into financially risky options during a crisis, such as taking out predatory payday loans or high-interest credit card debt.481 Lack of access to liquid assets can also make it harder to afford food and rent.482 June 2020 census data showed that, among households where a job was lost during the COVID-19 pandemic, 31 percent of African American households lacked sufficient food in the prior week, compared to 12 percent of white households.483 The data also showed that, compared to white renters, African American renters were less likely

to have paid their rent in the previous month and more likely to predict that they would not be able to make their next rent payment.<sup>484</sup> In a May 2020 survey, African American respondents were more than twice as likely as white respondents to report missing a credit card, utility, internet, rent, mortgage, or other "important payment" since the beginning of the pandemic.<sup>485</sup> African

American families with liquid assets also use them up more rapidly than white families during a crisis.<sup>486</sup> African American families with emergency savings at the start of the pandemic were twice as likely as white households to have needed to use them by May, and more than twice as likely to have already spent at least a quarter of their savings.<sup>487</sup>

#### **Discriminatory Tax Structures**

More than 50 years after the passage of the Civil Rights Act, many wealth-building policies still continue to heavily favor households that do not need help building wealth—mostly wealthy, predominantly white households—while doing little or nothing for low-wealth African American households, among other households of color.<sup>488</sup> The largest and most powerful of these programs operate through the U.S. tax code.<sup>489</sup>

These federal tax programs overwhelmingly favor building the wealth of the wealthy, and has contributed to the extreme rise in overall wealth inequality over the past several decades.<sup>490</sup> Tax policies have drastically different impacts on African American and white families and many were created during a time when African American families paid into the system without having the same legal rights to live, work, marry, vote, or receive an education as their white peers.<sup>491</sup>

The modern income tax system traces its roots to the Revenue Act of 1913, which instituted a progressive income tax system where tax rates increase as income increases but did not envision African Americans as taxpayers at all.<sup>492</sup> Even as African Americans eventually paid into the tax system after amendments and several new laws, they were unable to reap its benefits.<sup>493</sup> Today, African Americans are paying more in taxes than their white peers because U.S. tax laws were designed with white Americans in mind.<sup>494</sup>

The federal government has spent more than \$8 trillion in the past twenty years on tax programs to help families build long-term wealth through saving for retirement,

Although the Internal Revenue Service does not collect race or ethnicity data, recent research indicates that the overwhelming amount of the federal tax benefits goes to white households at all income levels.

buying a home, starting a business, or accessing higher education.<sup>495</sup> This spending has resulted in the typical millionaire receiving about \$145,000 in public tax benefits to grow their wealth, while working families get a total average of \$174.<sup>496</sup>

Although the Internal Revenue Service does not collect race or ethnicity data, recent research indicates that overwhelming amount of the spending the federal government does through the tax code goes to white households at all income levels.<sup>497</sup> Examining individuals earning \$103,466 or more reveals that white people accounted for 79 percent of those respective filers, while African American people accounted for only six percent.<sup>498</sup> These individuals also took greater advantage of high-value tax benefits, which cost the government hundreds of billions of dollars each year. For example, in 2012, white people made up 83 percent of the residents in the ZIP codes with the highest percentage of tax returns reporting capital gains and mortgage interest deductions.<sup>499</sup> But, African Americans made up just three percent of residents in the ZIP codes reporting the highest rates of capital gains income and six percent of residents in ZIP codes reporting the highest rates of mortgage interest deductions. These two tax programs together cost the federal government more than \$100 billion during that year.<sup>500</sup>

#### California

In California, Los Angeles provides a stark version of nationwide racial disparities in liquid assets accessible during a crisis.<sup>501</sup> A 2014 study of the Los Angeles metro area found that the median value of liquid assets for U.S.-African American households was \$200, compared to \$110,000 for white households.<sup>502</sup>

While 91.6 percent of white households had some kind of liquid asset, only 62.3 percent of U.S.-born African American households did. Further, 90.1 percent of white households had a checking account, versus 68.1 percent of U.S.-born African American households.<sup>503</sup> In addition, 71.9 percent of white households had a savings account, versus 55.5 percent of U.S.-born African American households.<sup>504</sup> While 40.7 percent of white households had stocks, mutual funds, or investment trusts, only 21.5 percent of U.S.-born African American households did.<sup>505</sup> Finally, 63.6 percent of white households had an individual retirement account or private annuity, versus 37.9 percent of U.S.-born African American households.<sup>506</sup>

# Percent of households in Los Angeles that hold some type of liquid asset (2014)



In California, many current state tax policies benefit Californians with higher incomes and wealth—predominantly white and Asian Californians—and give little to no benefits to other Californians, including African American Californians.<sup>507</sup> For example, many tax benefits are only available to those who opt to itemize their tax deductions, and people who itemize tend to have higher incomes. California's four largest personal income tax deductions provide most benefits to families with incomes over \$100,000, while providing nearly no benefits to those with incomes below \$20,000.<sup>508</sup> African American Californians are more likely to have low incomes, so they are less likely to benefit from these tax breaks.<sup>509</sup>

California's several tax benefits for homeowners, including the deductions for mortgage interest and property taxes, are also unlikely to benefit African American Californians who have been blocked from homeownership through racist policies and practices, often lack funds for a down payment, and have lower value homes due to residential segregation and racist appraisal practices.<sup>510</sup>

## **VI.** Conclusion

The legacy of slavery continues to reach into the lives of African Americans today. For hundreds of years, the American government at the federal, state, and local levels has systematically prevented African American communities from building, maintaining, and passing on wealth due to the racial hierarchy established to maintain enslavement.

Segregation, racial terror, harmful racist neglect, and other atrocities in nearly every sector of civil society

have inflicted harms, which cascade over a lifetime and compound over generations. As a result, African Americans today experience a large and persistent wealth gap when compared to white Americans. Addressing this persistent racial wealth gap means undoing long-standing institutional arrangements that have kept African American households from building and growing wealth at the same rate as white households to the present day.

## Endnotes

<sup>1</sup> Hicks et al., <u>Still Running Up the Down</u> <u>Escalator: How Narratives Shape Our</u> <u>Understanding of Racial Wealth Inequality</u> (2021) (as of Mar. 18, 2022).

<sup>2</sup> Ibid.

<sup>3</sup>Fed. Reserve Bank of Cleveland, <u>What is</u> <u>Behind the Persistence of the Racial Wealth</u> Gap? (Feb. 28, 2019) (as of Mar. 18, 2022).

<sup>4</sup>U.S. Federal Reserve, <u>Disparities in Wealth</u> <u>by Race and Ethnicity in the 2019 Survey</u> of Consumer Finances (Sept. 28, 202

 <sup>5</sup>Darity Jr. and Mullen, From Here to Equality: Reparations for African Americans in the 21st Century (2020) pp.
 35-38 (hereafter From Here to Equality).

<sup>6</sup>*Id.* at p. 36.

<sup>7</sup>Ibid.

<sup>8</sup> Darity Jr. et al, <u>A Subaltern Middle Class:</u> <u>The Case of the Missing Black Bourgeoisie</u> in Contemporary Economic Policy (Jul. 2020), 12476, p. 499.

<sup>9</sup>*Ibid.* 

<sup>10</sup>Fenaba et al., <u>Disparate Recoveries:</u> <u>Wealth, Race, and the Working Class After</u> <u>the Great Recession</u> in Annals of the American Academy of Political and Social Science (May 2021), , p. 182.

<sup>11</sup>Williamson, <u>Closing the racial wealth gap</u> requires heavy progressive taxation of wealth (Dec. 9, 2020), (as of Mar. 18, 2022).

<sup>12</sup>Kent, et al., <u>Has Wealth Inequality in</u> <u>America Changed over Time? Here are Key</u> <u>Statistics</u> (Dec. 2, 2020), Federal Reserve Bank of St. Louis, (as of January 19, 2022).

<sup>13</sup> Fed. Reserve Bank of Cleveland, What is Behind the Persistence of the Racial Wealth Gap?, supra.

<sup>14</sup>Dereroncourt, et al. <u>*The racial wealth*</u> <u>gap, 1860-2020</u> (Mar. 8, 2021), (as of Mar. 18, 2022), pp. 1-2.

<sup>15</sup>From Here to Equality, *supra*, p. 36; Kocchar and Fry, <u>Wealth inequality</u> <u>has widened along racial, ethnic lines,</u> <u>since the end of the Great Recession</u> (Dec. 12, 2014), (as of Mar. 18, 2022).

<sup>16</sup> Maddox, What are assets? The building blocks of wealth for individuals and

 $\frac{\text{profits for businesses}}{h}$  (Nov. 7, 2021),  $\frac{h}{2}$  (as of Mar. 18, 2022).

 $\frac{17}{-}$ Ibid.

<sup>18</sup> Darity, Jr., <u>The True Cost of Closing the</u> <u>Racial Wealth Gap</u>. The New York Times (Apr. 30, 2021), (as of Mar. 18, 2022).

 $\frac{19}{-}$ Ibid.

 $\frac{20}{2}$  From Here to Equality, *supra*, p. 257.  $\frac{21}{2}$  *Ibid*.

<sup>22</sup>Darity Jr., The True Cost of Closing the Racial Wealth Gap, supra.

 $\frac{23}{2}$ Ibid.

<sup>24</sup>Hicks et al. Still Running Up the Down Escalator, supra.

 $\frac{25}{2}$  *Id.* at p. 8.

 $\frac{26}{2}$  *Id.* at p. 6.

27

 <sup>28</sup> Bhutta, et al., <u>Disparities in Wealth by</u> <u>Race and Ethnicity in the 2019 Survey</u> of <u>Consumer Finances</u> (2020), <u>FEDS</u> <u>Notes</u> 2020-09-28-2, Board of Governors of the Federal Reserve System (U.S.) (Bhutta et al.) (as of Apr. 10, 2022).

 $\frac{29}{2}$ Ibid.

<sup>30</sup> From Here to Equality, *supra*, p. 36.
 <sup>31</sup> Ibid.

32 For the most recent studies that present documentation about the absence of any significant difference in Black and white saving behavior after controlling for household income, see Borgo, Ethnic and Racial Disparities in Saving Behavior in Journal of Economic Inequality, no. 17 (2019) pp. 253–83; and Traub et al., The Asset Value of Whiteness: Understanding the Racial Wealth Gap (Feb. 6, 2017. See also Gittleman and Wolff, Racial Differences in Patterns of Wealth Accumulation in Journal of Human Resources 39, no. 1 (2004) pp. 193-227; and Chiteji and Hamilton, Family Connections and the Black-White Wealth Gap among Middle-Class Families in Review of Black Political Economy 30, no. 1 (2002) pp. 9-28.

<sup>33</sup> What We Get Wrong About Closing the Racial Wealth Gap, supra.

#### $\frac{34}{-}$ Ibid.

<sup>35</sup>Traub et al., *Asset Value of Whiteness*, *supra*, pp. 7–8.

<sup>36</sup>See Mason, Race, Culture, and Skill: Interracial Wage Differentials among African Americans, Latinos, and Whites in Review of Black Political Economy 25, no. 3 (1997) pp. 5-39; Mangino, Race to College: The 'Reverse Gap' in Race and Social Problems, no. 2 (Dec, 2010) pp. 164-78; Mangino, Why Do Whites and the Rich Have Less Need for Education? in American Journal of Economics and Sociology 71, no. 3 (Jul. 2012) pp. 562–602; Mangino, The Negative Effects of Privilege on Educational Attainment: Gender, Race, Class, and the Bachelor's Degree in Social Science Quarterly 95, no. 3 (Sept. 2014) pp. 760-84.

 $\frac{37}{2}$  From Here to Equality, at p. 36.  $\frac{38}{2}$  *Ibid* 

<sup>39</sup> An Equity Profile of the Los Angeles Region (2017), Policy Link and the Program for Environmental and Regional Equity, (as of Mar. 18, 2022); De La Cruz-Viesca et al., <u>The Color of Wealth in Los Angeles</u> (2016), Fed. Reserve Bank of San Francisco, (as of Mar. 18, 2022); Ong, et al., <u>Race, Ethnicity, And Income</u> <u>Segregation in Los Angeles</u> (Jun. 24, 2016), UCLA Center for Neighborhood Knowledge, (as of Mar. 18, 2022).

 $\frac{40}{2}$ Ibid.

<sup>41</sup>De La Cruz-Viesca et al., *The Color of Wealth in Los Angeles, supra*, p.40.

 <sup>42</sup> See <u>Racial Wealth Divide Snapshot: Women</u> and the Racial Wealth Divide (Mar. 29, 2018), Prosperity Now, (as of Mar. 18, 2022).

 $\frac{43}{2}$ Ibid.

<sup>44</sup>See Gould et al, <u>What is the gender pay</u> <u>gap and is it real?</u>, Economic Policy Institute, (as of Apr. 14, 2022).

<sup>45</sup> Racial Wealth Divide Snapshot, supra.

 <sup>46</sup>See, e.g., Struvyen, <u>Black Womenomics:</u> <u>Investing in the Underinvested</u> (Mar.
 9, 2021), Goldman Sachs, <u>, p. 27</u> (as of Jan. 19, 2022); Pettit, review of McDermott, Mass Incarceration and the Myth of Black Progress (2012), https://www.journals.uchicago. edu/doi/abs/10.1086/672363?journalCode=ajs (as of Mar. 18, 2022).

 <sup>47</sup>Ibid. See also Kop, <u>Why the U.S. underreports the economic gaps between</u> <u>white and African Americans</u> (Jun. 3, 2020) (as of Mar. 18, 2022).

<sup>48</sup> Pettit, review of McDermott, Mass Incarceration and the Myth of Black Progress, *supra*.

 <sup>49</sup>Gramlich, <u>The gap between the number of</u> <u>blacks and whites in prison is shrinking</u> (Apr. 30, 2019), (as of Mar. 18, 2022)

<sup>50</sup>Darity Jr., What We Get Wrong About Closing the Racial Wealth Gap, supra.

<sup>51</sup>Struvyen, <u>Black Womenomics: Investing in</u> <u>the Underinvested</u> (Mar. 9, 2021), Goldman Sachs, p.6, (as of Jan. 19, 2022)

<sup>52</sup> Racial Wealth Divide Snapshot: Women and the Racial Wealth Divide, supra.

 <sup>53</sup>Struyven, Black Womenomics: Investing in the Underinvested, supra; Richard, <u>The</u> <u>Wealth Gap for Women of Color</u> (2014), Center for Global Policy Solutions Fact Sheet, (as of Mar. 18, 2022).

<sup>54</sup>Black Womenomics: Investing in the Underinvested, supra.

 $\frac{55}{-}$ Ibid.

 $\frac{56}{-}$ Ibid.

<sup>57</sup> Richard, The Wealth Gap for Women of Color, supra.

<sup>58</sup>Brown, <u>The Intersection and Accumulation</u> of Racial and Gender Inequality: Black <u>Women's Wealth Trajectories</u> (Jun. 2011) in The Review of Black Political Economy, (as of Mar. 18, 2022).

<sup>59</sup> Racial Wealth Divide Snapshot: Women and the Racial Wealth Divide, supra.

#### $\frac{60}{2}$ Ibid.

<sup>61</sup><u>Race: Total Population</u>. Explore census data. (n.d.). Retrieved February 22, 2022, from.

 <sup>62</sup>Tamir, <u>Key Findings About Black</u> <u>Immigrants in the U.S</u>. (Jan. 27, 2022), Pew Research Center, (as of Mar. 18, 2022).
 <sup>63</sup>Ibid.  $\frac{64}{-}$ Ibid.

 $\frac{65}{-}$ Ibid.

#### $\frac{66}{2}$ Place of Birth for the Foreign-Born

<u>Population in California</u>, Explore census data, (n.d.), Retrieved March 18, 2022, from

<sup>67</sup>Hicks et al. Still Running Up the Down Escalator, supra.

 <sup>68</sup> Hamilton, Immigration and the Remaking of Black America (May 2019).
 <sup>69</sup> Ibid.

<sup>70</sup> De La Cruz-Viesca et al., *The Color of Wealth in Los Angeles, supra*. Aja, et al, *The Color of Wealth in Miami* (Feb. 2019), (as of Mar. 18, 2022). Munoz, et al., *The Color of Wealth in Boston*, Fed. Reserve of Boston, (as of Mar. 18, 2022); Kijakazi, et al, *The Color of Wealth in the Nation's Capital* (Nov. 2016), (as of Mar 18, 2022).
 <sup>71</sup> Ibid.

 <sup>72</sup> Roderick Graham, "<u>Reflections on the ADOS</u> <u>Movement</u>," Medium, March 3, 2019, (as of Mar. 18, 2022).

 $^{73}$  De La Cruz-Viesca et al., The Color of Wealth in Los Angeles, supra.

- $\frac{74}{-}$ Ibid.
- $\frac{75}{-}$ Ibid.
- $\frac{76}{-}$ Ibid.

<sup>77</sup> Ibid.

<sup>78</sup> Ibid.

<sup>79</sup>*Ibid*.

<sup>80</sup>From Here to Equality, *supra*, p. 258.

<sup>81</sup>*Id.* at pp. 253-258.

<sup>82</sup>*Id.* at p. 253.

<sup>83</sup>*Ibid.* 

 <sup>84</sup> <u>California Task Force to Study and Develop</u> <u>Reparation Proposals for African Americans</u> (October 12, 2021), Testimony of Thomas Craemer (as of October 25, 2021).

<sup>85</sup> Craemer, Estimating Slavery Reparations: Present Value Comparisons of Historical Multigenerational Reparations Policies in Social Science Quarterly 96, no. 2 (June 2015), pp. 639–55.

 $\frac{86}{-}$ Ibid.

<sup>87</sup> Marketti, Estimated Present Value of Income Directed during Slavery in America, Wealth of Races, pp. 107–24.

<sup>88</sup>From Here to Equality, *supra*, p. 254.

 $\frac{89}{-}$  Id. at p. 255.

90 *Ibid*.

<sup>91</sup>*Id.* at pp. 255-56.

<sup>92</sup> Darity Jr. and Mullen, <u>William A. Darity</u> and Kristen Mullen on Direct Payments to Close the Racial Wealth Gap (May 18, 2021), (as of Mar. 18, 2022).

<sup>93</sup>Allen, Homesteading and Property Rights; or, 'How the West Was Really Won in The Journal of Law and Economics 34, no. 1, (1991).

94 Ibid.

 <sup>95</sup>Library of Congress, <u>Homestead Act:</u>
 <u>Primary Documents in American History</u> (n.d.), (as of Mar. 18, 2022).

<sup>96</sup>University of Nebraska-Lincoln -Center for Great Plains Studies, <u>Homesteading</u> (n.d.), (as of Mar. 18, 2022).

<sup>97</sup> National Park Service, <u>A History of</u> <u>African Americans in California: Farming</u> (n.d.) (as of Mar. 18, 2022).

<sup>98</sup> Ibid.; National Park Service, <u>African</u> <u>American Homesteaders in the Great</u> <u>Plains</u>, (as of Mar. 18, 2022).

<sup>99</sup> Merritt, <u>Land and the roots of African-</u> <u>American Poverty</u> (Mar. 11, 2016), AEON (as of Mar. 18, 2022).

<sup>100</sup>*Ibid*.

<sup>101</sup>National Park Service, African American Homesteaders in the Great Plains, supra.

<sup>102</sup> From Here to Equality, *supra*, p. 255;
 Darity, Jr., *Forty Acres and a Mule in the 21st* <u>Century</u> in Social Science Quarterly 89,
 No. 3 (Sept. 2008), (as of Mar. 18, 2022).

<sup>103</sup> Brown, Forty Acres and a Mule: How the First Reparations for Slavery Ended in Betrayal (Apr. 15, 2021), (as of Mar. 18, 2022).

<sup>104</sup>Darity Jr., Forty Acres and a Mule in the 21st Century, supra.

 <sup>105</sup> Brown, Forty Acres and a Mule: How the First Reparations for Slavery
 Ended in Betrayal, supra.
 <sup>106</sup> Ibid. <sup>107</sup> Darity Jr., Forty Acres and a Mule in the 21st Century, supra.

 $\frac{108}{10}$  Ibid.

 $\frac{109}{10}$  Ibid.

<u>110</u> Ibid.

<sup>111</sup> Brown, Forty Acres and a Mule: How the First Reparations for Slavery Ended in Betrayal, supra.

<sup>112</sup>Darity Jr., Forty Acres and a Mule in the 21st Century, supra.

<u>113</u>*Ibid*.

 $\frac{114}{2}$ Ibid.

<u>115</u> Ibid.

<sup>116</sup>From Here to Equality, *supra*, p. 256.

 <sup>117</sup> Darity Jr. and Mullen, <u>William A. Darity</u> and Kristen Mullen on Direct Payments to Close the Racial Wealth Gap (May 18, 2021), (as of Mar. 18, 2022).

<u>118</u>*Ibid.* 

<u>119</u>*Ibid.* 

120 *Ibid*.

Logan and Darity Jr., <u>Look What Has</u>
 <u>Been Taken from African Americans</u>
 (Sept. 9, 2021), as of Mar. 18, 2022).

<sup>122</sup> Taylor Jr., <u>Breonna Taylor's death and</u> racist police violence highlight the danger of gentrification (Jul. 22, 2020), (as of Mar. 18, 2022); Maharawal, <u>Gentrification</u> and the Security State (2017), (as of Mar. 18, 2022); Fayyad, <u>The Criminalization</u> of Gentrifying Neighborhoods (2017), The Atlantic, (as of Mar. 18, 2022).

<sup>123</sup> Darity Jr. and Mullen, William A. Darity and Kristen Mullen on Direct Payments to Close the Racial Wealth Gap, supra.

 $\frac{124}{2}$ Ibid.

<sup>125</sup> Public Broadcasting Service, <u>Go</u>
 <u>Deeper: Where Race Lives</u> (n.d.),
 (as of Mar. 18, 2022) (PBS).

<sup>126</sup>See Chapter 5

<sup>127</sup> PBS, supra.

<sup>128</sup> Rothstein, The Color of Law: A Forgotten History of How Our Government Segregated America (2017), at preface.

<sup>129</sup> Center for American Progress, Systematic Inequality, Displacement, *Exclusion, and Segregation* (Aug. 7, 2019), (as of Mar. 18, 2022) (Center for American Progress).

<sup>130</sup> Darity Jr. and Mullen, *William A. Darity* and Kristen Mullen on Direct Payments to Close the Racial Wealth Gap, supra.

 $\frac{131}{2}$  PBS, supra.

 $\frac{132}{2}$  Center for American Progress, *supra*.

<sup>133</sup>See Chapter 5

<sup>134</sup> Darity Jr. and Mullen, William A. Darity and Kristen Mullen on Direct Payments to Close the Racial Wealth Gap, supra.

#### $\frac{135}{2}$ Ibid.

<sup>136</sup>Daniel, Dispossession: Discrimination against African American Farmers in the Age of Civil Rights (2015), pp. 3-5.

<sup>137</sup> *Id.* at p. 34.
<sup>138</sup> *Id.* at p. 28.
<sup>139</sup> *Id.* at pp. 23, 124, 130.
<sup>140</sup> *Id.* at p. 28.
<sup>141</sup> *Id.* at p. 220.
<sup>142</sup> *Id.* at p. 216.
<sup>143</sup> *Id.* at p. 216.
<sup>144</sup> *Id.* at p. 217.
<sup>145</sup> *Id.* at pp. 218, 229.
<sup>146</sup> *Id.* at pp. 219, 229.
<sup>147</sup> *Id.* at p. 232.

<sup>148</sup> California Task Force to Study and Develop Reparation Proposals for African Americans (October 13, 2021), <u>Testimony of Lawrence</u> Lucas, (as of February 10, 2022).

 <sup>149</sup>Moore, <u>Plight of Black Farmers in the</u> <u>Context of USDA Farm Loan Programs:</u> <u>A Research Agenda for the Future</u> (December 16, 2013), Professional Agriculture Workers Journal, (as of February 10, 2022)

 $\frac{150}{2}$  Ibid.

 $\frac{151}{2}$ Ibid.

 $\frac{152}{2}$  Ibid.

<sup>153</sup> Healy, <u>A Windfall for Minority Farmers</u>
<u>Divides Rural America</u> (May 22,
2021) The New York Times (as
of February 10, 2022)

<sup>154</sup>Darity Jr. and Mullen, William A. Darity and Kristen Mullen on Direct Payments to Close the Racial Wealth Gap, supra.

<sup>155</sup>See Chapter 11

<u>156</u>*Ibid*.

 <sup>157</sup> Matos and. Hodge, <u>The Chains of Slavery</u> <u>Still Exist in Mass Incarceration</u> (Jun. 17, 2021), Vera Institute for Justice, (as of Mar. 18, 2022).

#### <u>158</u> Ibid.

 <sup>159</sup>See generally Pope, Mass Incarceration, Convict Leasing, and the Thirteenth Amendment: A Revisionist Account (2019)
 94 N.Y.U. L. Rev. 1465, 1467.

<sup>160</sup>Oshinsky, <u>Worse Than Slavery</u> (Apr. 28, 1996 (as of Mar. 18, 2022).

<sup>161</sup>*Ibid.*; see also Blackmon, Slavery by Another Name: The Re-Enslavement of African Americans from the Civil War to World War II (2008); Cardon, *"Less Than Mayhem": Louisiana's Convict Lease*, 1865-1901 (2017) Louisiana History: The Journal of the Louisiana Historical Association vol. 58, no. 4 pp. 417–441 <> (as of July 23, 2021).

<sup>162</sup> Hinton and Cook, <u>The Mass</u> <u>Criminalization of African Americans: A</u> <u>Historical Overview</u> in Annual Review of Criminology (Jan. 2021), (as of Mar. 18, 2022) (citing Oshinsky, *supra*).

<sup>163</sup>Matos and Hodge, <u>The Chains of Slavery</u>
 <u>Still Exist in Mass Incarceration</u> (Jun. 17, 2021), Vera, (as of Apr. 14, 2022).

164 See Chapter 11.

<u>165</u> Ibid.

<sup>166</sup> Pettit, Mass Incarceration and the Myth of Black Progress, *supra*.

<sup>167</sup> Harper et al, <u>Debt, Incarceration, and</u> <u>Re-entry: A Scoping Review</u> (Aug. 11, 2020), (as of Mar. 18, 2022).

 <sup>168</sup> Kent, <u>Has Wealth Inequality in America</u> <u>Changed over Time? Here are Key Statistics</u>
 (December 02, 2020) Federal Reserve
 Bank of St. Louis (as of January 19, 2022)

<sup>169</sup>Zaw et al., <u>Race, Wealth and</u> <u>Incarceration: Results from the</u> <u>Longitudinal Survey of Youth</u> (Feb. 15, 2016), (as of Mar. 18, 2022).

#### Chapter 13 - The Wealth Gap

<sup>170</sup>Krivo & Peterson, <u>Extremely disadvan-</u> taged neighborhoods and urban crime
(1996) Social Forces 75(2): pp.
619-647 < [as of July 29, 2021].</li>

 <sup>171</sup>Krivo et al., <u>Segregation, racial structure,</u> and neighborhood violent crime
 (2009) American Journal of
 Sociology vol. 114, no. 6 pp.
 1765–1802 > (as of July 29, 2021).

<sup>172</sup> Federal Bureau of Investigation – Criminal (as of April 14, 2022); <u>U.S. Census Bureau,</u> <u>Quick Facts</u>: U.S. > (as of July 22, 2021).

 <sup>173</sup> Pierson et al., <u>A large scale analysis of</u> racial disparities in police stops across the United States (2017) Nature Human Behavior 4, 736–745 (2020) (as of July 27, 2021).

<sup>174</sup>*Ibid*.

<u>175</u>*Ibid.* 

<sup>176</sup>Zaw et al., *Race*, *Wealth and Incarceration*, *supra*.

#### <sup>177</sup>*Ibid*.

<sup>178</sup>Harper et al., Debt, Incarceration, and Re-entry: A Scoping Review, supra.

<sup>179</sup>See Chapter 11

<sup>180</sup>Western and Wildeman, *The Black Family and Mass Incarceration* in the The Annals of the American Academy of Political and Social Science. 2009;621(1):221–242. doi: 10.1177/0002716208324850.

#### <u>181</u> Ibid.

<sup>182</sup>Z aw, et al., *Race, Wealth and Incarceration, supra.* 

#### <sup>183</sup>*Ibid.*

<sup>184</sup> Federal Bureau of Prisons, <u>Federal</u>
 <u>Prison Industries Program Details</u>,
 < (as of July 27, 2021).</li>

<sup>185</sup>Hoffer, <u>Involuntary Servitude: How Prison</u> <u>Labor is Modern Day Slavery</u> (Feb. 3, 2022), Harvard Political Review, (as of Mar. 18, 2022) (Hoffer).

<u>186</u> Ibid.

<sup>187</sup>*Ibid.* 

<sup>188</sup>Hoffer, supra; see also Stygar, Thinking Outside the Box: A Point-Based System of Reintegration for California's Inmate Firefighters (2020) 56 Cal. Western L.Rev. 455, 456 (Stygar).

<sup>189</sup>Hoffer, Involuntary Servitude: How Prison Labor is Modern Day Slavery, supra.

 $\frac{190}{2}$  Ibid.

<sup>191</sup> See Prison Policy Initiative, <u>State and</u> <u>federal prison wage policies and sourc-</u> <u>ing information</u> (Apr. 10, 2017) (as of Mar. 18, 2022); Salay, <u>When in pris-</u> <u>on, the costs are steep and the pay close</u> to nothing, (as of Mar. 18, 2022).

 <sup>192</sup> Benns, <u>American Slavery, Reinvented</u> (Sept. 21, 2015), The Atlantic, (as of Mar. 24, 2022).

<sup>193</sup>See generally Hammad, Shackled to Economic Appeal: How Prison Labor Facilitates Modern Slavery While Perpetuating Poverty in Black Communities (2019) 26 Va. J. Soc. Pol'y & L. 65, 66.

<sup>194</sup>Zaw et al., *Race, Wealth and Incarceration, supra.* 

 $\frac{196}{2}$ Ibid.

<sup>197</sup> Bivens et al., *Reforming Unemployment Insurance: Stabilizing a System in Crisis and Laying the Foundation for Equity* (June 2021), Economic Policy Inst., https:// files.epi.org/uploads/Reforming-Unemployment-Insurance.pdf (as of Apr. 2, 2022) (Bivens et al.).

<sup>198</sup>U.S. Dep't. of Agriculture, <u>Economic</u> Linkages: Supplemental Nutrition Assistance Program (SNAP) Linkages with the General Economy (Aug. 20, 2019) Economic Research Service, < (as of Aug. 19, 2021).</p>

<sup>199</sup>Brantley et al., <u>Association of Work</u> <u>Requirements with Supplemental</u> <u>Nutrition Assistance Program Participation</u> <u>by Race/Ethnicity and Disability</u> <u>Status, 2013-2017</u> (June 2020), (as of Mar. 18, 2022), pp. 7-8.

<sup>200</sup> Katznelson, When Affirmative Action
 Was White: An Untold History of
 Racial Inequality in Twentieth Century America (2005), p. 23 (When
 Affirmative Action Was White).
 <sup>201</sup> Ibid.

202 *Ibid*.

203 Ibid.
204 Id. at pp. 42-43.
205 Ibid.
206 Ibid.
207 Ibid.
208 Ibid.
209 Ibid.
210 Ibid.
211 Ibid.
212 Ibid.

<sup>213</sup>Trotter, Jr., From a Raw Deal to a New Deal?: 1929-1945 (The Young Oxford History of African Americans) (Apr. 5, 1996); To Make Our World Anew Volume Two: A History of African Americans since 1880 (2005), edited by Robin D.G. Kelley and Earl Lewis, New York: Oxford University Press, 2005, p. 131.

<sup>214</sup>Onkst, First a Negro... Incidentally a Veteran: Black World War Two Veterans and the G.I. Bill of Rights in the Deep South, 1944-1948 (Spring 1998) 31(3) in Journal of Social History 517, 518 (Onkst); Knutson, <u>Legislation to Resolve GI Bill Racial Inequities Introduced by</u> <u>Democrats</u> (Nov. 11, 2021) Axios (as of Jan. 18, 2022); Katznelson, Fear Itself: The New Deal and the Origins of Our Time (Mar. 1, 2013), p. 368 (Fear Itself).

<sup>215</sup> Fear Itself, *supra*, pp. 368, 620; see also When Affirmative Action Was White, *supra*, pp. 113–41; Mettler, Soldiers to Citizens: The G.I. Bill and the Making of the Greatest Generation (New York: Oxford University Press, 2007); Frydl, The G.I. Bill (Cambridge: Cambridge University Press, 2011).

<sup>216</sup> Fear Itself, *supra*, pp. 368, 620; see also
When Affirmative Action Was White, *supra*, pp. 113–41; Mettler, Soldiers
to Citizens: The G.I. Bill and the
Making of the Greatest Generation
(New York: Oxford University Press,
2007); Frydl, The G.I. Bill (Cambridge:
Cambridge University Press, 2011).

<sup>217</sup>Onkst, *supra*, p. 520; Rothenberg, Race, Class, and Gender in the United States: An Integrated Study (2007) p. 46.

 $<sup>\</sup>frac{195}{--}$  Ibid.

<sup>218</sup> Katznelson & Mettler, <u>On Race and</u> <u>Policy History: A Dialogue About the G.I.</u> <u>Bill</u> (Sept. 2008) 6(3) American Political Science Association 519, pp. 521, 522, 530, available at (as of Jan. 18, 2022).

 <sup>219</sup> Fording et al., <u>Race and the Local Politics</u> of <u>Punishment in the New World of</u> <u>Welfare</u> (March 2011) 116(5) American Journal of Sociology 1610, 1614, 1650 (Fording et al.). (available on JSTOR)

<sup>220</sup>*Id.* at p. 1614.

 $\frac{221}{2}$ Bivens et al, *supra*, at pp. 2, 10.

<sup>222</sup>*Id.* at p. 22.

 $\frac{223}{2}$  *Id.* at p. 10.

<sup>224</sup> Costa, Government Accountability Office, Management Report: Preliminary Information on Potential Racial and Ethnic Disparities in the Receipt of Unemployment Insurance Benefits during the COVID-19 Pandemic (June 17, 2021), Management Report: Preliminary Information on Potential Racial and Ethnic Disparities in the Receipt of Unemployment Insurance Benefits during the COVID-19 Pandemic (as of Apr. 4, 2022) p. 3 (GAO Report), p. 3.

<sup>225</sup>Brantley et al., *supra*, pp. 7-8.

226 Chen, Center for Retirement
 Research, Boston College, <u>Did the</u>
 <u>Stimulus Checks Help People with</u>
 <u>Unexpected Expenses?</u> (Aug. 2021),
 p. 3(as of Mar. 18, 2022).

<sup>227</sup>Bivens, et al., supra.

<sup>228</sup>U.S. Dep't of the Treasury, <u>Paycheck</u>
<u>Protection Program</u>, (as of Aug. 19, 2021).> (as of Aug. 19, 2021).

229 Trisi & Saenz, Center on Budget and Policy Priorities, <u>Economic Security</u>
<u>Programs Reduce Overall Poverty, Racial and</u>
<u>Ethnic Inequities</u> (July 1, 2021); Holtzblatt
& Karpman, Tax Policy Center, Urban
Inst., <u>Who Did Not Get the Economic Impact</u>
<u>Payments by Mid-to-Late May, and Why?</u>
(July 2020) (as of Apr. 2, 2022) p. 2.

<sup>230</sup> Price & Pugh, <u>The Next Pandemic</u>
<u>Stimulus Bill Must be Race-Conscious</u> (Sept. 4, 2020), Time, (as of Aug. 19, 2021).

<sup>231</sup> Plerhoples, <u>Correcting Past Mistakes: PPP</u> Loans and Black-Owned Small <u>Businesses</u> (Feb. 25, 2021) Am. Const. Society (as of Aug. 19, 2021).

<sup>232</sup> Mills & Battisto, Federal Reserve Bank of New York, *Double Jeopardy: Covid-19's Concentrated Health and Wealth Effects in Black Communities* (Aug. 2020) p. 5-6.

<sup>233</sup> *Ibid.*; Plerhoples, *supra*.

 $\frac{234}{2}$  Ibid.

<sup>235</sup>California Task Force to Study and Develop Reparation Proposals for African Americans (October 12, 2021), <u>Testimony of Rucker C. Johnson,</u> <u>Ph.D</u>, (as of October 25, 2021).

 $\frac{236}{10}$  From Here to Equality, *supra*, p. 37.

 <sup>237</sup> Jones, <u>The racial wealth gap: How African</u> Americans have been shortchanged out of the materials to build wealth (Feb 13, 2017), (as of Apr. 15, 2022).

<sup>238</sup> California Task Force to Study and Develop Reparation Proposals for African Americans (October 12, 2021), <u>Testimony of Rucker C. Johnson,</u> <u>Ph.D</u>, (as of October 25, 2021).

<sup>239</sup>Herring and Henderson, <u>Wealth</u> <u>Inequality in Black and White: Cultural</u> <u>and Structural Sources of the Racial</u> <u>Wealth Gap</u> (Feb. 2016), (as of Mar.
18, 2022) (Herring and Henderson)
p. 6; Chiteji, <u>The Racial Wealth Gap</u> <u>and the Borrower's Dilemma</u>, p. 356.

<sup>240</sup>Zaw, et al., <u>Women, Race and Wealth</u>, Samuel DuBois Cook Center on Social Equity, Duke University and Insight Center for Community Economic Development (2017).

241 Pat S., Income vs. Wealth – What's the Difference? (Sept. 14, 2021), Money Crasher, LLC, (as of Mar. 18, 2022).

 $\frac{242}{2}$ Ibid.

<sup>243</sup>Zaw, et al.., *Women, Race and Wealth, supra.* 

 $\frac{244}{2}$ Ibid.

<sup>245</sup>See Chapter 10

<sup>246</sup>From Here to Equality, *supra*, at p. 256.

 $\frac{247}{2}$ Ibid.

 $\frac{248}{2}$ Ibid.

<sup>249</sup> Swinton, "Racial Inequality and Reparations," in America, Wealth of Races (1990), pp. 153–62.

 $\frac{250}{2}$ Ibid.

 $\frac{251}{2}$ Ibid.

252 Ibid.

<sup>253</sup>Herring and Henderson, *supra*.

<sup>254</sup> Barsky et al., Accounting for the Black-White Wealth Gap: A Nonparametric Approach (Sep. 2002) in Journal of the American Statistical Association Vol. 97, No. 459, pp. 663-673.

<sup>255</sup>*Ibid.* 

<sup>256</sup>*Ibid.* 

<sup>257</sup> Darity Jr., What We Get Wrong About Closing the Racial Wealth Gap, supra. <sup>258</sup> Ibid.

<sup>259</sup> Blau and Graham, <u>Black/White</u> <u>Differences in Wealth and Asset Composition</u> (Mar. 1989), available at (as of Mar. 24, 2022) (Blau and Graham) ; Darity Jr., What We Get Wrong About Closing the Racial Wealth Gap, supra; From Here to Equality, supra; Herring and Henderson, supra.

<sup>260</sup>Darity Jr. What We Get Wrong About Closing the Racial Wealth Gap, supra, p. 22.

<sup>261</sup> Blau and Graham, supra; Darity Jr., What We Get Wrong About Closing the Racial Wealth Gap, supra; From Here to Equality, supra; Herring and Henderson, supra.

<sup>262</sup> Baradaran, The Color of Money: Black Banks and the Racial Wealth Gap (2017), Cambridge, Harvard University Press (Baradaran) p. 30.

<u><sup>263</sup></u>Ibid.

<sup>264</sup> Garrett-Scott, Banking on Freedom: Black Women in U.S. Finance before the New Deal (2019), Columbia University Press, New York (Garrett-Scott) pp. 13, 31.

#### $\frac{265}{2}$ Ibid.

<sup>266</sup>*Id.* at pp. 15-18.

<u><sup>267</sup></u>*Ibid.* 

268 *Ibid*.

269 Ibid.

<sup>270</sup> Ibid.

<sup>271</sup>*Ibid*.

#### Chapter 13 - The Wealth Gap

<sup>272</sup>*Ibid.* 

<sup>273</sup>*Ibid*.

 274 Washington, U.S. National Archive, *The Freedman Savings and Trust Company* in Federal Records and African American History Summer 1997, Vol. 29, No. 2, (as of Apr. 2, 2022) (Washington).

 $\frac{275}{2}$ Ibid.

<sup>276</sup>Garrett-Scott, *supra*, at p. 25

<sup>277</sup>Baradaran, *supra*, at p. 25.

<sup>278</sup>Washington, *supra*; Garrett-Scott, *supra*, at p. 25; Baradaran, *supra*, at pp. 25-26.

<sup>279</sup>Washington, *supra*.

<sup>280</sup>*Ibid.*; Garrett-Scott, *supra*, at p. 25

<sup>281</sup>Washington, supra.

282 Baradaran, supra, at p. 25-26.

<sup>283</sup>*Ibid*.

<sup>284</sup>*Ibid*.

 $\frac{285}{10}$  Id. at p. 30.

<sup>286</sup>*Ibid.*; Garrett-Scott, *supra*, pp. 13, 31; U.S. Senate 43rd Congress, 2nd Session (1874); Du Bois, Black Reconstruction in America 1860-1880 (1998), New York: Simon and Schuster, p. 600; Gilbert, The Comptroller of the Currency and the Freedman Savings Bank, The Journal of Negro History Vol. LVII No. 2, (as of Apr. 2, 2022) p. 131 (Gilbert); The U.S. Treasury. The Freedman's Savings Bank: Good Intentions Were Not Not Enough, A Noble Experiment Goes Awry (as of Apr. 2, 2022); Nier III, The Shadow of Credit: The Historical Origins of Facial Predatory Lending and ITS Impact Upon African American Wealth Accumulation, U. Pa. J.L. & Soc. Change, p. 151-152 (2008) (as of Apr. 2, 2022) (Nier III)

 <sup>287</sup> Célérier and Tak, <u>The Impact of</u> Financial Inclusion on Minorities: Evidence from the Freedman's Savings Bank (Aug. 1, 2021), (as of Mar. 18, 2022) p. 32.

<sup>288</sup> Baradaran, *supra*, p. 30; Garrett-Scott, *supra* note 302, p. 13; p. 31; U.S. Senate
43rd Congress, 2nd Session (1874); Du
Bois, *Black Reconstruction in America 1860-1880*, New York: Simon and Schuster,
1998, p. 600; Gilbert, "The Comptroller of the Currency," p. 131; and the U.S.

Treasury <u>The Freedman's Savings Bank:</u> <u>Good Intentions Were Not Not Enough, A</u> <u>Noble Experiment Goes Awry</u>, (as of Apr. 2, 2022) state that the number of depositors were 61,144 at the time of its closing.

<sup>289</sup>Harmon et al., Negro as a Business
Man, p. 54; Baradaran, *supra*, p. 296;
Seder and Burrell, Getting It Together, 9.

 $\frac{290}{1}$  Nier III, "The Shadow of Credit," p. 151-152.

 $\frac{291}{2}$  Id. at p. 154.

 $\frac{292}{2}$ Ibid.

 $\frac{293}{2}$ Ibid.

 $\frac{294}{2}$ Ibid.

<sup>295</sup> Ammons, <u>The Evolution of Black-Owned</u> <u>Banks in the United States between the</u> <u>1880s and 1990s</u> in Journal of Black
Studies 26, no. 4 (1996), , p. 471; Neal and Walsh, <u>The Potential and Limit of</u> <u>Black-Owned Banks</u> (Mar. 18, 2020), Urban Institute, (as of Mar. 18, 2022).

<sup>296</sup> Baradaran, *supra*, at p. 87.

 <sup>297</sup>Wangman and Wooldridge, <u>32 Black-owned banks and credit unions</u>, <u>sorted by state</u> (Feb 1, 2022), Business Insider, (as of Mar. 18, 2022).
 <sup>298</sup>Ibid.

<sup>299</sup>Baradaran, *supra*, at pp. 88-89

300 *Ibid.* 301 *Ibid.* 

- 302 Ibid.
   303 Ibid.
   304 Ibid.
- 305 *Ibid*.
- <u>306</u>*Ibid.*
- <u>307</u>*Ibid.*
- 308 *Ibid*.
- <u>309</u>*Ibid.*
- 310 *Ibid*.

<sup>311</sup>Lindsay, *The Negro in Banking* (Apr. 1929) in The Journal of Negro History, The University of Chicago Press on behalf of the Association for the Study of African American Life and History, p. 195.

 $\frac{312}{2}$  Ibid.

<sup>313</sup> National Park Service, African American Homesteaders in the Great Plains, supra; National Park Service, A History of Black Americans in California: Farming, supra.
<sup>314</sup> Ibid.

 $\frac{315}{2}$  *Ibid*.

<sup>316</sup> National Park Service, *African American Homesteaders in the Great Plains, supra.* 

<sup>317</sup>Cal. Prison Industrial Authority, <u>Catalogue</u> (as of Nov. 24, 2021).

 $\frac{318}{-}$ Ibid.

<sup>319</sup> *Ibid*.

<sup>320</sup> Stygar, supra.

 $\frac{321}{2}$ Ibid.

322 Hutchful, <u>The Racial Wealth Gap:</u>
 What California Can Do About a
 Long-Standing Obstacle to Shared
 Prosperity, California Budget &
 Policy Center (2018), p. 8.

 $\frac{323}{2}$ Ibid.

<u>324</u> Ibid.

 $\frac{325}{2}$ Ibid.

<sup>326</sup>De La Cruz-Viesca et al., *The Color of Wealth in Los Angeles* (2016), *supra*.

<sup>327</sup> Bell et al., Cal. Policy Lab, <u>An Analysis</u> of Unemployment Insurance Claims in California During the COVID-19 <u>Pandemic</u> (Dec. 2020) p. 10.

<sup>328</sup>*Id.* at p. 22.

<sup>329</sup>CalFresh, <u>The Able-Bodied Adults</u>
 <u>Without Dependents Time Limit</u>, Cal. Dept.
 Social Services, (as of Aug. 19, 2021).

<sup>330</sup> Botts & Rodriguez-Delgado, <u>Getting</u>
 <u>Food Stamps to Poor Californians is</u>
 <u>Surprisingly Difficult</u> (Sept. 2, 2019)
 CalMatters (as of Aug. 19, 2021).

 $\frac{331}{2}$ Ibid.

<sup>332</sup>U.S. Census Bureau, <u>Quick Facts:</u> <u>California</u> (Jul. 1, 2021), (as of April 15, 2022).

<sup>333</sup>Tan & Danielson, *The CalFresh Food Assistance Program* (Sept. 2020) Public
 Policy Inst. Cal. (as of Aug. 19, 2021).

<sup>334</sup> Ramos-Yamamato, <u>Not Enough to</u> Eat: California Black and Latinx Children Need Policymakers to Act (Sept. 2020) Cal. Budget & Policy Center (as of Aug. 19, 2021).

<sup>335</sup>Ong et al., Latino Policy & Politics
Initiative, Univ. Cal. Los Angeles,
Disparities in the Distribution
of Paycheck Protection Program
Funds Between Majority-White
Neighborhoods and Neighborhoods
of Color in California (Dec. 2020)
p. 5. (as of Apr. 9, 2022).

 <sup>336</sup> Morel et al., <u>Rampant Racial Disparities</u> <u>Plagued How Billions of Dollars in PPP</u> <u>Loans Were Distributed in the U.S.</u>
 (May 1, 2021 (as of Aug. 19, 2021).

<sup>337</sup> Lohrentz, <u>The Impact of Proposition 209</u> on California's <u>MWBE's</u> (Jan. 2015), Equal Justice Society, (as of Mar. 18, 2022), p.2. <sup>338</sup> Ibid.

\_\_\_\_*1010.* 

339 *Id.* at pp. 2-3.

 $\frac{340}{2}$ Ibid.

 $\frac{341}{2}$ Ibid.

 $\frac{342}{2}$  Id. at p. 3.

<sup>343</sup>See, e.g., Bowser, The Black Middle Class: Social Mobility – And Vulnerability (2006) p. 120.

<sup>344</sup> Bortz, <u>How to Unlock the Equity in Your</u>
<u>Home</u> (Aug. 10, 2021), N.Y.
Times, (as of Aug. 18, 2021).

 <sup>345</sup> Killewald and Bryan, "Does Your Home <u>Make You Wealthy</u>?" RSF: The Russell Sage Foundation Journal of the Social Sciences 2, no. 6, 2016, (Killewald and Bryan) p. 111.
 <sup>346</sup> Ibid.

<sup>347</sup>U.S. Dep't of Housing and Urban Development, <u>*Glossary of Terms to*</u> *Affordable Housing*, (as of Apr. 2, 2022).

<sup>348</sup>Killewald and Bryan, *supra*, p. 111.

 $\frac{349}{2}$ Ibid.

 <sup>350</sup> Darity Jr., What We Get Wrong About Closing the Racial Wealth Gap, supra, p. 13.
 <sup>351</sup> Ibid.

352 *Ibid*.

<sup>353</sup> Moss et al., <u>The Black-White Wealth Gap</u>
 <u>Left Black Households More</u>
 Vulnerable (2020), Brookings.

<sup>354</sup> <u>The State of The Nation's Housing 2020</u>
 (2020), Joint Center for Housing
 Studies of Harvard University, (as

of Apr. 2, 2022) (The State of the Nation's Housing 2020) p. 26.

<sup>355</sup> Flitter, <u>This is What Racism Sounds Like in</u> <u>the Banking Industry</u> (Dec. 11, 2019), N.Y. Times, (as of Apr. 2, 2022), Bartlett et al., <u>Consumer Lending Discrimination in</u> <u>the Fintech Era</u> (Jun. 2019), Nat. Bureau of Econ. Research, (as of Apr. 2, 2022).

<sup>356</sup>From Here to Equality, *supra*, p. 35.

<sup>357</sup> Tippett et al., Center for Global Policy Solutions, <u>Beyond Broke: Why Closing the</u> <u>Racial Wealth Gap is a Priority for National</u> <u>Economic Security</u> (May 2014), <u>https://</u> globalpolicysolutions.org/wp-content/ uploads/2014/04/BeyondBroke\_Exec\_ <u>Summary.pdf</u> (as of Mar. 18, 2022) p. 4. <u>358</u> Ibid.

<u>359</u>*Ibid.* 

 <sup>360</sup>Immergluck, <u>Racial Justice and the</u> <u>Mortgage Market: Recommendations</u> to Racial Justice and the Mortgage <u>Market: Recommendations to the Biden</u> <u>Administration Regarding the Future of</u> <u>the GSEs</u> (May 21, 2021), (as of April 15, 2022) p. 4 (Immergluck).

 $\frac{361}{2}$ Ibid.

<sup>362</sup>Herring and Henderson, *supra*, p. 15.

<sup>363</sup> Manhertz, <u>Home Value Disparities</u>
Between Races are Shrinking, but Remain Very
<u>Wide</u> (Dec. 19, 2020) Zillow (as of Aug. 18, 2020); Perry et al., Brookings Inst., <u>The</u>
Devaluation of Assets in Black Neighborhoods:
<u>The Case of Residential Property</u> (Nov. 2018), (as of Apr. 2, 2022) pp. 13-15.

<sup>364</sup> See Redfin, <u>Black Homeowners Earned</u>
\$59,000 in Home Equity in 2020, <u>Compared with \$50,000 for White</u> <u>Homeowners</u> (Mar. 9, 2021) PR Newswire (as of Aug. 18, 2021).

365 Perry et al., *supra*, p. 3.

 <sup>366</sup> Flitter, Where State Farm Sees "a Lot Of Fraud" Black Customers See Discrimination (Mar. 18, 2022) N.Y.
 Times, (as of Apr. 2, 2022).

#### $\frac{367}{2}$ Ibid.

<sup>368</sup> The State of The Nation's Housing 2020 (2020), Joint Center for Housing Studies of Harvard University, (as of Apr. 2, 2022) (The State of the Nation's Housing 2020) p. 26. <sup>369</sup> Darity, Jr., What We Get Wrong About Closing the Racial Wealth Gap, supra, p. 14.

<sup>370</sup>The State of the Nation's Housing2020, *supra*, p. 5.

<sup>371</sup> Murillo, <u>Black Homeowners Are Being Left</u> out of the Mortgage Refinance <u>Boom</u> (June 9, 2021) Money (as of Aug. 18, 2021) (Murillo).

<sup>372</sup> Campisi, <u>Pandemic Remodeling Boom</u>
 <u>Looming</u>? Cash-Out Refinancing
 Hits 13-Year High (Dec. 4, 2020)
 Forbes (as of Aug 18, 2021).

<sup>373</sup>Murillo, supra.

<sup>374</sup>What We Get Wrong About Closing the Racial Wealth Gap, supra.

<sup>375</sup>*Ibid*.

<sup>376</sup>Blau and Graham, *supra*, p. 16.

<sup>377</sup>Oliver and Shapiro, *supra*.

<sup>378</sup> Conley, Being Black, Living in the Red: Race, Wealth and Social Policy in America, Berkeley, CA, University of California Press, 1999 (Conley, Being Black), p. 41

 <sup>379</sup> Brimmer, Income, Wealth, and Investment Behavior in the Black
 *Community*, The American Economic Review 78, 2, 1988, pp. 151-155.

<sup>380</sup>Herring and Henderson, *supra*, p. 8.

 <sup>381</sup>Keister, <u>Race and Wealth Inequality: The</u> <u>Impact of Racial Differences in Asset</u>
 <u>Ownership on the Distribution of Household</u>
 <u>Wealth</u>, Social Science Research 29,
 2000, p. 499, (as of Apr. 10, 2022).

 <sup>382</sup> Brooks, <u>The Retirement Crisis Facing</u> <u>African Americans</u> (Dec. 11, 2020), U.S. News, (as of Apr. 2, 2022).
 <sup>383</sup> Ibid.

<sup>384</sup> Devaney, et al., <u>Asset Ownership by Black</u> and White Families in Financial Counseling and Planning 18, 1, 2007, p. 33-45, (as of April 10, 2022).

 <sup>385</sup> Florant, et al., <u>The case for accelerating</u> financial inclusion in Black communities (Feb. 25, 2020), (as of April 9, 2022).

<sup>386</sup>De La Cruz-Viesca et al, *supra*, p. 25.

<sup>387</sup>Akabas et al., <u>New BPC Survey Shows</u> <u>Americans Need Better Ways to Save for</u> <u>Emergencies</u> (Feb. 25, 2021) Bipartisan Policy Center (as of Aug. 18, 2021) (Akabas et al.); Hutchful, *supra*, p. 3

<sup>388</sup> Mills and Zhang, Urban Inst., <u>Social</u> <u>Support Networks and their Effects</u> <u>on Hardship Avoidance Among</u> <u>Low-Income Households (Dec.</u> 2013), (as of Apr. 10, 2022).

<sup>389</sup>Liu and Parilla, <u>New Data Shows Small</u> <u>Businesses in Communities of Color Had</u> <u>Unequal Access to Federal COVID-19 Relief</u> (Sept. 17, 2020) Brookings Inst. (as of Aug. 18, 2021) (Liu and Parilla).

<sup>390</sup>Akabas, et al., supra.

<sup>391</sup>Lerner, <u>Accessing Home Equity in a Tight</u>
 <u>Economy Has Become Problematic</u> (June 11, 2020) Wash. Post (as of Aug. 18, 2021).

<sup>392</sup>Akabas et al., supra.

<sup>393</sup> E.g., Gladstone et al., <u>Racial Inequities</u> <u>Have Long Existed in the Economy. A New</u> <u>BPC Survey Shows COVID-19 Made Them</u> <u>Worse.</u> (June 22, 2020) Bipartisan Policy Center (as of Aug. 18, 2021); Hardy and Logan, Hamilton Proj., Brookings Inst., Racial Economic Inequality Amid the Covid-19 Crisis (Aug. 2020) p. 2; Murillo, *supra*; Choe, <u>Stocks are Soaring</u>, <u>and Most Black People are Missing Out</u> (Oct. 12, 2020) AP News (as of Aug. 18, 2021); Bivens et al., *supra*; Brantley et al., *supra*; Bhutta et al., *supra*; Akabas et al., *supra*; Liu and Parilla, *supra*.

<sup>394</sup> Bhutta et al., supra.; Schuetz,
 <u>Rethinking Homeownership Incentives to</u>
 <u>Shrink the Racial Wealth Gap</u> (Mar. 15,
 2022), Shelterforce (as of Apr. 15, 2022).

 <sup>395</sup> Houser, Cleveland 19 News, <u>Cleveland</u> Man Alleges Racial Profiling after Bank <u>Refuses to Cash Check, Calls 9-1-1</u> (Dec.
 19, 2018), (as of Apr. 10, 2022).

 <sup>396</sup> Johnson, WSB-TV 2, <u>Doctor Says Bank</u> <u>Employees Called Police on Him While</u> <u>Trying to Open Account</u> (Sept. 13, 2019), (as of Apr. 10, 2022).

<sup>397</sup> Shammas, Miami New Times, <u>Another</u> <u>Black South Floridian Says Wells Fargo</u> <u>Refused to Cash His Check, Called Cops</u> (Aug. 9, 2018), (as of Apr. 10, 2022).

<sup>398</sup> Logan, The Independent, <u>A Florida</u> Bank Refused to Cash an Elderly Black Woman's Check. Then They 

 Called the Police on Her (July 28, 2018), (as of Apr. 10, 2022).

 399

 Hutchful, supra, p. 2.

 400

 E.g., Bhutta et al., supra.

 401

 Choe, supra; Bhutta et al., supra.

 402

 Choe, supra.

 403

 Bhutta et al., supra.

 404

 Ibid.

 405

 Ibid.

 406

 Ibid.

 408

 Ibid.

 409

 Akabas et al., supra.

<sup>410</sup> Perry and Romer, *To Expand the Economy, Invest in Black Businesses* (Dec. 31, 2020) Brookings
 Inst. (as of Apr. 10, 2022).

<sup>411</sup>Sobol, Factors Influencing Private Capital Accumulation on the 'Eve of Retirement' (1979) in Review of Economics and Statistics, LXI, pp. 585-593.

<sup>412</sup> Kroeger, Entrepreneurship and the Racial Wealth Gap: The Impact of Entrepreneurial Success or Failure on the Wealth Mobility of Black and White Families (2021), (Apr. 15, 2022); Lofstrom and Bates, African-Americans' Pursuit of Self-Employment (Nov. 2007), The Institute for the Study of Labor, (Apr. 15, 202

<sup>413</sup> Association for Enterprise
Opportunity, <u>The Tapestry of Black</u>
<u>Business Ownership in America:</u>
<u>Untapped Opportunities for Success</u>
(2017) (Association for Enterprise

Opportunity) p. 8, (as of Apr. 10, 2022). <sup>414</sup>*Ibid.* 

 $\frac{415}{10}$  Id at pp. 4, 8.

 $\frac{416}{1}$  Id. at p. 4.

 $\frac{417}{2}$  Ibid.

<sup>418</sup> Klein, <u>Bridging the Divide, How Business</u>
 <u>Ownership Can Help Close the Racial Wealth</u>
 <u>Gap</u> (2017) FIELD at the Aspen Institute
 (Klein) p. 6, (as of Apr. 10, 2022).

<sup>419</sup> Howard et al., <u>Banks & the Black</u> <u>Community, What Can Major Commercial</u> and Retail Banking Institutions Do to <u>Better Support Black Entrepreneurs and</u>

et al.), (as of Apr. 10, 2022). <sup>420</sup> De La Cruz-Viesca et al., The Color of Wealth in Los Angeles, supra, p. 41. 421 Howard et al., *supra*, pp. 1-3.  $\frac{422}{2}$  *Id.* at p. 3. <sup>423</sup>Association for Enterprise Opportunity, supra, pp. 24-28.  $\frac{424}{2}$  Ibid.  $\frac{425}{2}$  Ibid. 426 Ibid. 427 Ibid. 428 Klein, *supra*, p. 7.  $\frac{429}{2}$  *Id.* at p. 8.  $\frac{430}{2}$  Ibid.  $\frac{431}{1}$  Id. at p. 9.  $\frac{432}{2}$  Ibid. <sup>433</sup>From Here to Equality, *supra*, p.41. <sup>434</sup> Perry and Romer, *supra*. <sup>435</sup>Washington, *Covid-19 Has Had a* Disproportionate Impact on Black Small Businesses (June 3, 2021) Forbes  $\frac{436}{1}$ Ibid.  $\frac{437}{2}$  Ibid. 438 *Ibid*. 439 *Ibid*. 440 Ibid.  $\frac{441}{2}$  *Ibid*. 442 Davalos et al., Cal. Budget & Policy Center, California's 17 Million Renters Face Housing Instability and Inequity Before and After COVID-19 (Jan.

Businesses in the U.S.? (2020) (Howard

2021) pp. 6-7, (as of Apr. 10, 2022).

Hutchful, supra, at pp..6-7; Passy, <u>Black</u>
 <u>Homeownership Rate Hits Lowest Level</u>
 <u>Since the 1960s – That's Unlikely to Change</u>
 <u>in Pandemic Year 2</u> (Mar. 10, 2021)
 MarketWatch (as of Aug. 20, 2021).

<sup>445</sup>Hutchful, *supra*, at 6.

<sup>446</sup>Shapiro et al., Institute on Assets and Social Policy, <u>The Roots of the Widening</u> <u>Racial Wealth Gap: Explaining the</u> <u>Black-White Economic Divide</u> (Feb. 2013), p.3-4, (as of Apr. 11, 2022).  $\frac{447}{2}$ Hutchful, *supra*, at 6.

448 Ong. et al, supra.

 <sup>449</sup>Hall et al., <u>Neighborhood Foreclosures</u>, <u>Racial/Ethnic Transitions</u>, and <u>Residential</u> <u>Segregation</u> (2015) in Am. Sociol.
 Rev., (as of Apr. 15, 2022).

<sup>450</sup> De La Cruz-Viesca, et al., *supra*, pp.29-30.

451 Ali, *supra*, p. 67.

<sup>452</sup>De La Cruz-Viesca, et al., *supra*, p. 36.

453 *Id.* at p. 41.

<sup>454</sup> Blau and Graham, supra, Conley, supra; Killewald, Return to Being Black, Living in the Red: A Race Gap in Wealth That Goes Beyond Social Origins (2013) in Demography 50 (Killewald);
Weller et al., Simulating How Large Policy Proposal Affect the Black-White Wealth Gap (2021) in Journal of Economics, Race and Policy, p. 2.

<sup>455</sup>Toney & Robertson, <u>Intergenerational</u> <u>Economic Mobility and the Racial Wealth</u> <u>Gap</u>, American Economic Association Papers & Proceedings, Vol. 111 (May, 2021), pp. 206-10, (as of Apr. 11, 2022).

 $\frac{456}{2}$ Ibid.

 $\frac{457}{2}$ Ibid.

<sup>458</sup> Dalton Conley, Decomposing the Black-White Wealth Gap: The Role of Parental Resources, Inheritance, and Investment Dynamics (2001) in Sociological Inquiry 71, 1 (Conley, Decomposing), pp. 39-66.; Herring and Henderson, supra; Weller et al., supra.

<sup>459</sup>Blau and Graham, *supra*; Conley, *Being Black*, *supra*;; Killewald, *supra*; Weller et al., *supra*.

<sup>460</sup> McKernan, et al.. *Private Transfers, Race, and Wealth* (2012), The
Urban Institute, Report 5,
p.11, (as of Apr. 2, 2022).

 <sup>461</sup>Rosalsky, NPR Planet Money, <u>There is</u> <u>Growing Segregation in Millenial Wealth</u> (Apr. 27, 2021), (as of Apr. 11, 2022).

 $\frac{462}{2}$ *Ibid.* 

 $\frac{463}{2}$  Ibid.

 $\frac{464}{2}$  Ibid.

465 *Ibid*.

<sup>466</sup>Blau and Graham, supra, Conley, <u>Being</u> <u>Black, supra; Killewald, Return to Being</u> <u>Black, Living in the Red: A Race Gap in</u> <u>Wealth That Goes Beyond Social Origins</u>
(2013) in Demography 50 (Killewald),
(as of Apr. 11, 2022); Weller et al., <u>Simulating How Large Policy Proposal</u> <u>Affect the Black-White Wealth Gap</u> (2021)
in Journal of Economics, Race and Policy, p. 2, (as of Apr. 11, 2022).

<sup>467</sup>Rosalsky, *supra*.

468 *Ibid*.

 $\frac{469}{-}$ Ibid.

<sup>470</sup>Whitehead, Center for Poverty Research, Univ. Cal. Davis, <u>Support</u> from Family Does Not Replace the Social <u>Safety Net</u> (Apr. 2016) (Whitehead)
p. 1, (as of Apr. 11, 2022).

471 Ibid.al

<sup>472</sup>Airgood-Obrycki et al., <u>Making the Rent</u> <u>Household Spending Strategies During the</u> <u>COVID-19 Pandemic</u> (Jan. 2022), Joint
Center for Housing Studies Harvard
University, (as of Apr. 15, 2022); Airgood-Obrycki et al., <u>Renters' Responses to</u> <u>Financial Stress During the Pandemic</u> (Apr. 2021), Joint Center for Housing Studies
Harvard University, (as of Apr. 15, 2022).

<sup>473</sup>Whitehead, *supra*, at p. 1.

 $\frac{474}{-}$  Id. at p. 2.

 $\frac{475}{2}$ Bhutta et al., *supra*.

476 Hutchful, supra, at p. 2.

<sup>477</sup> Gladstone et al., *supra*; Weidrich and Newville, <u>Vulnerability in the Face of</u> <u>Economic Uncertainty: Key Findings from the</u> <u>2019 Prosperity Now Scorecard</u> (Jan. 2019), Prosperity Now, (as of Apr. 15, 2022).
<sup>478</sup> Bhutta et al., *supra*.
<sup>479</sup> Akabas et al., *supra*.
<sup>480</sup> Gladstone et al., *supra*.

<sup>481</sup>Watkins and Akabas, <u>The Savings Gap</u> for Americans is A Key Component of Economic Stability – Here's what we Can

Do About it (July 1, 2020) Bipartisan Policy Center (as of Aug. 18, 2021). 482 *Ibid.* 483 Gladstone et al., supra. 484 *Ibid*. 485 *Ibid*. 486 *Ibid*. 487 *Ibid*. 488 Asante-Muhammad, et al, Corporation for Economic Development, *The Ever-Growing Gap*: Without Change, African-American and Latino Families Won't Match White Wealth for Centuries (Aug. 2016), (as of Apr. 11, 2022). 489 *Ibid*. 490 *Ibid*. <sup>491</sup>Brown, The Whiteness of Wealth (2021) p. 16. <sup>492</sup>*Id.* at p. 12. <sup>493</sup>*Id.* at pp. 14-15. <sup>494</sup>*Id.* at p. 21. <sup>495</sup>Asante-Muhammad, et al, *supra*, p. 17.  $\frac{496}{2}$ Ibid. <sup>497</sup>*Id.* at p. 18. 498 *Ibid*. 499 *Ibid*. 500 *Ibid*. <sup>501</sup>De La Cruz-Viesca et al., The Color of Wealth in Los Angeles, supra. 502 Id. at p. 38. 503 Id. at p. 25. 504 *Ibid*. 505 *Id.* at p. 6. 506 Id. at pp. 6-7. 507 *Ibid*. 508 *Id.* at p. 4. 509 *Ibid*.  $\frac{510}{10}$  Id. at p. 5.